# THIRD AMENDED AND RESTATED GOVERNANCE AGREEMENT CONCERNING DEVELOPMENT OF THE RAIL PROJECTS

by and among

NORTHWEST INDIANA REGIONAL DEVELOPMENT AUTHORITY, NORTHERN INDIANA COMMUTER TRANSPORTATION DISTRICT,

and

INDIANA FINANCE AUTHORITY

# TABLE OF CONTENTS

ARTICLE I.		DEFINITIONS OF TERMS AND LIST OF ACRONYMS USED	3
ARTICLE II.		CONDITIONS PRECEDENT	16
ARTICLE III.		PROJECT MANAGEMENT AND COORDINATION	16
ARTICLE IV.		PROPERTY INTERESTS AND ACTIVITIES	20
ARTICLE V.		FINANCING AND FUNDING	22
ARTICLE VI.		NICTD OPERATIONS COVENANTS	24
ARTICLE VII.		DESIGN, CONSTRUCTION AND DEVELOPMENT OF THE RAIL PROJECTS	26
ARTICLE VIII.		OPERATIONS AND MAINTENANCE	35
ARTICLE IX.		INSURANCE AND BONDING	38
ARTICLE X.		DISPUTE RESOLUTION	40
ARTICLE XI.		ADVERSE CONDITIONS / ENHANCED OVERSIGHT/STEP-IN RIGHTS	41
ARTICLE XII.		RESTORATION	46
ARTICLE XIII.		TERMINATION	46
ARTICLE XIV.		THIRD PARTY AGREEMENTS	46
ARTICLE XV		GENERAL MATTERS	47
		<u>EXHIBITS</u>	
Exhibit 1	Dou	able Track Project	
Exhibit 2	West Lake Corridor Project		
Exhibit 3A	Dou	ıble Track Project Cost Budget	
Exhibit 3B	We	st Lake Corridor Project Cost Budget	
Exhibit 4A	pit 4A Double Track Project Schedule		
Exhibit 4B Wes		st Lake Corridor Project Schedule	
Exhibit 5A	Dou	ible Track Financing Plan	

Exhibit 5B	West Lake Corridor Financing Plan
Exhibit 6	West Lake Operating Deficit Calculation
Exhibit 7	Agreed Form of Calculation for Revenue Coverage and Debt Coverage
Exhibit 8	West Lake Deficit Cost Submission Form
Exhibit 9	SSI

# THIRD AMENDED AND RESTATED GOVERNANCE AGREEMENT CONCERNING DEVELOPMENT OF THE RAIL PROJECTS

This Third Amended and Restated Governance Agreement Concerning Development of the Rail Projects (this "Governance Agreement"), effective as of \_\_\_\_\_\_, 2022 (the "Effective Date"), is made by and among the Northwest Indiana Regional Development Authority ("RDA"), the Northern Indiana Commuter Transportation District ("NICTD") and the Indiana Finance Authority ("IFA") for the purpose of amending and restating the Second Amended and Restated Governance Agreement Concerning Development of the Rail Projects (the "Original Second Amended Governance Agreement"), effective as of July 1, 2020 (the "Effective Date of the Second Amended Governance Agreement"), as amended and supplemented by the First Amendment to Second Amended and Restated Governance Agreement Concerning Development of the Rail Projects, dated as of February 3, 2021 (the "First Amendment" and together with the Original Second Amended Governance Agreement, the "Second Amended Governance Agreement"), which amended and restated the Amended and Restated Governance Agreement Concerning Development of the Rail Projects (the "Amended Governance Agreement"), effective as of June 24, 2019 (the "Effective Date of the Amended Governance Agreement"), which amended and restated the Governance Agreement Concerning Development of the Rail Projects (the "Original Governance Agreement"), effective as of August 31, 2018 (the "Effective Date of the Original Governance Agreement"), each by and among RDA, NICTD, and IFA. RDA, NICTD, and IFA are sometimes referred to individually herein as a "Party" or collectively as the "Parties."

#### Recitals

- A. WHEREAS, the South Shore Line is a commuter train service between South Bend, Indiana and Millennium Station in Chicago, Illinois;
- B. WHEREAS, NICTD is a commuter transportation district and a municipal corporation formed pursuant to IC 8-5-15;
- C. WHEREAS, NICTD has owned and operated the South Shore Line for nearly forty years and has experience in developing capital rail projects for the South Shore Line and in requesting, obtaining and managing federal grant funds through the Federal Transit Administration:
- D. WHEREAS, RDA is a separate body corporate and politic formed pursuant to IC 36-7.5 to foster development in northwest Indiana and has since its inception fostered economic development throughout northwest Indiana and has been vested pursuant to IC 36-7.5-4-18 with certain powers and duties with respect to the projects hereinafter described;
- E. WHEREAS, IFA is a body politic and corporate formed pursuant to IC 5-1.2-3-1 to oversee State debt issuance and provide effective financing solutions to facilitate State, local and business investment in Indiana and has participated in numerous important economic development activities throughout Indiana;
- F. WHEREAS, the RDA and NICTD have developed and advanced plans for improvements to the South Shore Line, specifically the Double Track Project and the West Lake

Corridor Project more fully described herein (collectively, the "Rail Projects"), which will improve the South Shore Line and foster economic development throughout northwest and north central Indiana by shortening commute times, providing new rail service to areas not serviced by rail transportation to Chicago, enhancing the economic and community development of the northwest Indiana region, and increasing the convenience to the areas served;

- G. WHEREAS, the RDA has determined that the execution and delivery of this Governance Agreement is authorized pursuant to the provisions of IC 5-1.3 and that providing assistance to the Rail Projects in the manner herein provided serves a public purpose for which the RDA was created;
- H. WHEREAS, it is expected that the Rail Projects will also encourage business investment in and near the station areas, provide faster, less expensive and reliable transportation, will relieve existing and anticipated traffic congestion, reduce vehicular emissions, reduce commuting costs and attract and retain families and younger residents in northwest and north central Indiana:
- I. WHEREAS, the Rail Projects will provide faster travel times, will accommodate more trains to improve service during rush hour travel and will be safer to operate;
- J. WHEREAS, the RDA, IFA and NICTD have identified and obtained committed local and State funding for the Rail Projects;
- K. WHEREAS, NICTD has entered into Full Funding Grant Agreements (the "FFGAs") with the Federal Transit Administration ("FTA") for both Rail Projects on behalf of the RDA and the State and, in its capacity as the CIG grantee and as the signatory party to the FFGAs, will maintain continuing control of the assets receiving the FTA CIG funding, subject to the terms herein and the Leases;
- L. WHEREAS, any obligations of the RDA or the IFA herein shall not constitute a debt of the State or any municipality, or a pledge of the full faith and credit thereof, under the Constitution of the State;
- M. WHEREAS, additional funding for the Rail Projects has been made available by the State of Indiana pursuant to the provisions of IC 8-14-14.3-5 and IC 5-13-10.5-20;
- N. WHEREAS, NICTD has agreed to be responsible for certain additional costs related to the construction of the West Lake Corridor Project as set forth herein;
- O. WHEREAS, RDA has agreed to provide additional funding to be dedicated to the design and construction of the West Lake Corridor Project;
- P. WHEREAS, since the date of the First Amendment, \$68,556,588 of additional funding has been allocated to the Rail Projects by the FTA through the American Rescue Plan Act of 2021;

- Q. WHEREAS, since the date of the First Amendment, the General Assembly appropriated certain funds to the IFA to be utilized for the Rail Projects in lieu of the proceeds of the IFA Appropriation Bonds (as defined in the Revenue Trust Fund Agreement);
- R. WHEREAS, bids for the design-bid-build contract for the Double Track Project were received, but because the bids received exceeded the then effective Double Track Project Cost Budget they were rejected;
- S. WHEREAS, based on feedback from the initial and prospective respondents to the Double Track Project design-bid-build procurement, the Parties determined to reallocate contractual risk between NICTD and the Double Track Contractor for DT 1 with the goal of increasing competition and lowering bid amounts in a re-bid of DT 1;
- T. WHEREAS, NICTD proposed to pay for fifty percent (50%) of certain delay costs for excusable delays to the Critical Path and Project Milestones (each as described in DT 1) caused by utilities and other third-parties at no fault of the Double Track Contractor for DT 1;
- U. WHEREAS, as a condition for their consent to include in DT 1 the 50% cost sharing by NICTD for third-party excusable delays, IFA and RDA require that NICTD agree to bear and pay any of the 50% delay costs that exceed the Double Track Project Cost Budget;
- V. WHEREAS, on November 22, 2021, the Board of Trustees of NICTD adopted a resolution awarding DT 1 to Walsh/Herzog Joint Venture ("Walsh");
- W. WHEREAS, on December 21, 2021, NICTD and Walsh entered into DT1 in the amount of \$375,519,909.30, and DT1 included the provisions described above;
- X. WHEREAS, the Parties have determined that the Double Track Project Cost Budget needs to be increased to \$\_\_\_\_\_\_\_; and
- Y. WHEREAS, in connection with the foregoing, the Parties desire to amend and restate the Second Amended Governance Agreement, as set forth herein;
- NOW, THEREFORE, in consideration of the mutual agreements set forth herein, the Parties agree as follows:

#### ARTICLE I.

#### **DEFINITIONS OF TERMS AND LIST OF ACRONYMS USED**

- 1.1 **Advanced Double Track Delay Costs** means Double Track Delay Costs that initially were paid as Project Costs from funding sources as provided in the Financing Plan for the Double Track Project, which funding NICTD is required to pay and reimburse in accordance with Section 7.8.4 in the event and to the extent that the Cost Budget of the Double Track Project has been exceeded.
- 1.2 **Advanced WL Delay Costs** means WL Delay Costs that initially were paid as Project Costs from funding sources as provided in the Financing Plan for the West Lake Corridor

Project, which funding NICTD is required to pay and reimburse in accordance with Section 7.8.3 in the event and to the extent that the Cost Budget of the West Lake Corridor Project has been exceeded.

- 1.3 **Annual Report** means the Year End Performance Report prepared by NICTD and provided to IFA and RDA pursuant to <u>Section 8.7.</u>
  - 1.4 **Alternate(s)** has the meaning set forth in the WL Design-Build Contract.
- 1.5 **Availability Breach** means a failure to maintain the South Shore Line for use and availability in the manner provided in the Leases, unless such failure is due to a Force Majeure event.
- 1.6 **Availability Breach Notice** means notice by IFA or RDA of an Availability Breach.
- 1.7 **ARPA Funds** means funds allocated to the Rail Projects by the FTA through the American Rescue Plan Act of 2021.
- 1.8 **Availability Cure Period** means the 30-day period following an Availability Breach Notice or such longer period as the Parties may agree in writing.
- **Books and Records** means any and all documents, books, records, papers, or other information relating to the South Shore Line and/or the obligations herein, including (i) all design and construction documents, and operations and maintenance documents, including, but not limited to, drawings, specifications, submittals, subcontracts, subconsultant agreements, purchase orders, invoices, schedules, meeting minutes, budgets, forecasts, change orders, (ii) income statements, balance sheets, statements of cash flow and changes in financial position, operating income, expenses, capital expenditures and budgeted operating results relating to the Rail Projects, (iii) all budgets, certificates, claims, contract agreements, correspondence, data (including test data), documents, expert analyses, facts, files, information, investigations, materials, notices, plans, projections, proposals, records, reports, requests, samples, schedules, settlements, statements, studies, surveys, tests, test results, and (iv) with respect to all of the above, any information that is stored electronically or on computer-related media, including in any electronic document management system; provided however, that nothing in the Agreement shall require the disclosure by any Party of Books and Records protected by the attorney-client or other legal privilege based upon an opinion of counsel reasonably satisfactory to the other Party, and all disclosures shall be subject to the SSI Policy, if applicable.
- 1.10 **Bond Resolution** means Resolution No. 20-15 (adopted by NICTD on November 2, 2020), or such other bond resolution substantially similar in form and acceptable to IFA and RDA entered into or otherwise executed or adopted for the purpose of issuing and/or securing bonds or other Permitted NICTD Debt issued by NICTD as provided in <u>Section 6.3 and Section 6.4</u>.
  - 1.11 **Bond Trustee** means the bank serving as the trustee for the IFA Bonds.

- 1.12 **Business Plan** means the annual report prepared by NICTD detailing the plan for operations and maintenance for the forthcoming year and the five-year capital plan for the South Shore Line as provided in <u>Section 8.8</u>.
  - 1.13 **Capital Expenditures** has the meaning set forth in <u>Section 6.1(e)</u>.
- 1.14 **Construction Adverse Condition** means the occurrence of one or more of the conditions in Sections 11.1.1 11.1.9.
  - 1.15 **Construction Adverse Condition Notice** means a notice pursuant to <u>Section 11.2</u>.
- 1.16 **Construction Period Step-In Rights** means the rights of IFA and/or RDA to step in and assume design and construction of the Rail Projects pursuant to Section 11.4.
  - 1.17 **Consultation Period** has the meaning set forth in Section 11.6.2.
- 1.18 **Cost Budget** means the Project Costs and Project Contingency for design, construction, and development for completion of each of the Rail Projects, which are attached as <a href="Exhibit 3A"><u>Exhibit 3A</u></a> (Double Track Project Cost Budget) and <a href="Exhibit 3B"><u>Exhibit 3B</u></a> (West Lake Corridor Project Cost Budget), as applicable, which may be amended by the Parties.
  - 1.19 **CSS** means Chicago South Shore & South Bend Railroad.
- 1.20 **Debt Coverage** means the calculation of the debt service coverage ratio as provided in <u>Section 6.3</u> and <u>Exhibit 7</u> and certified by NICTD in a manner consistent with <u>Exhibit 7</u>, and subject to verification by the IFA and/or RDA in their discretion as provided in <u>Section 6.8</u>.
- 1.21 **Decision Making Framework Factors** are the factors to be considered during the decision making process set forth in Section 3.1.2.
- 1.22 **Deposit Trustee** means the bank serving as trustee under the Revenue Trust Fund Agreement.
- 1.23 **Design-Builder** means the entity with whom NICTD has entered the WL Design-Build Contract, dated October 29, 2020, and any successor or assignee entity thereto.
- 1.24 **Development Agreement** means an agreement that may be negotiated and executed by the Parties prior to the issuance of any bonds or other financing for the Rail Projects which may set forth additional terms and conditions for the Rail Projects, responsibilities of the Parties, means of addressing changed circumstances and the details of the financing plans.
- 1.25 **Double Track Contractors** means the entities with whom NICTD has entered into the contracts designated as DT 1, DT 2, DT 3, DT 4, DT 5, DT 6 and DT 7 to construct or provide goods and services for any portion of the Double Track Project, and any successor or assignee entities thereto, as well as the entity under contract to construct the parking garage in Michigan City pursuant to the development agreement between NICTD and Michigan City, and any successor or assignee entity thereto, but only to the extent the construction of such project is not otherwise covered under the DT 2 contract.

- 1.26 **Double Track Delay Costs** means, and is solely limited to, costs validly due and owing by NICTD to the Double Track Contractor for DT 1 pursuant to the fifty percent (50%) shared cost provisions in Section 12.5.6 (utility delays) and/or Section 12.9 (third-party agreement delays) of DT 1, as it may hereafter be amended or supplemented.
- 1.27 **Double Track Portion of the Maximum NICTD Excess Cost Contribution** means not more than \$15,000,000 of the Maximum NICTD Excess Cost Contribution, which is allocated to the Double Track Project as provided in <u>Section 7.8.2</u>.
- 1.28 **Double Track Project** means the addition of an east/west rail line for a portion of the South Shore Line from Gary to Michigan City and related improvements, all as more fully described in Exhibit 1.
- 1.29 **Draw Down Bonds** means the IFA Bonds issued for the Double Track Project pursuant to IC 5-13-10.5-20(d).
- 1.30 **DT 1** means the Construction Contract, dated December 21, 2021, between NICTD and a Double Track Contractor, with respect to that portion of the Double Track Project generally consisting of the track, platforms, structures, municipal utilities, select roadways and select surface parking lots.
- 1.31 **DT 2** means the Construction Contract, dated August 31, 2021, between NICTD and a Double Track Contractor, with respect to that portion of the Double Track Project generally consisting of the Michigan City garage and stationhouse and retail space.
- 1.32 **DT 3** means the Construction Contract, dated August 10, 2021, between NICTD and a Double Track Contractor, with respect to that portion of the Double Track Project generally consisting of the Gary (Miller) stationhouse and east parking lot.
- 1.33 **DT 4** means the Construction Contract, dated \_\_\_\_\_\_\_, 2021, between NICTD and a Double Track Contractor, with respect to that portion of the Double Track Project generally consisting of the construction of the south parking lot and improvements to Stagecoach Road.
- 1.34 **DT 5** means the Construction Contract, dated April 5, 2021, between NICTD and a Double Track Contractor, with respect to that portion of the Double Track Project generally consisting of the construction of the Dune Park extension of the east parking lot.
- 1.35 **DT 6** means the Goods and Services Contract, dated \_\_\_\_\_\_\_, 202\_, between NICTD and a Double Track Contractor, with respect to that portion of the Double Track Project generally consisting of the NICTD centralized traffic control/supervisory control and data acquisition and positive train control systems and associated systems modifications.
- 1.36 **DT 7** means the Construction Contract, dated August 11, 2021, between NICTD and a Double Track Contractor, with respect to that portion of the Double Track Project generally consisting of exclusion fencing.
  - 1.37 **Enhanced Oversight** means the activities provided for in <u>Section 11.2</u>.

- 1.38 **Equivalent Alternative Funding** means funding secured from the General Assembly in the manner provided in <u>Section 8.4.5</u> in an amount that is equal to or exceeds the immediately prior four quarters of Tax Credit Dollars, whether received by or provided to any Party or any other entity, in each case in any manner that is dedicated or made available for use to subsidize the West Lake Operating Deficit.
- 1.39 **Escrow Fund** means the escrow fund into which RDA is to deposit Tax Credit Dollars as provided in <u>Section 8.4.1</u>.
- 1.40 **Excess Construction Costs** means Project Costs in excess of the Cost Budget as of the Effective Date for completion of the applicable Rail Project.
- 1.41 **FFGA** means a full funding grant agreement executed with FTA for either Rail Project.
  - 1.42 **FFGA Approval** means execution of an FFGA by FTA.
- 1.43 **Final Completion**, with respect to the West Lake Corridor Project, has the meaning set forth in the WL Design-Build Contract, and with respect to the Double Track Project, means the latest date for Final Completion, as defined in DT 1, DT 2, DT 3, DT 4, DT 5, DT 6 or DT 7.
- 1.44 **Financing Plan** means the plan prepared by RDA and IFA, and agreed to by NICTD, that describes how NICTD, RDA, and IFA intend to finance and fund the State and local share of the capital costs of each of the Rail Projects, and the agreed sources and uses of the monetary contributions, grant funds, and bond proceeds for the Rail Projects, which shall be integrated into the FTA Financial Plan for each Rail Project prepared by NICTD as required by the FTA CIG Program. The respective Financing Plans are attached as <u>Exhibit 5A</u> (Double Track Financing Plan) and <u>Exhibit 5B</u> (West Lake Corridor Financing Plan).
  - 1.45 **FONSI** means a Finding of No Significant Impact signed by the FTA.
- 1.46 **Force Majeure** means an Act of God, fire, tornado, hurricane, flood, earthquake, explosion, war on American soil, terrorist acts, civil disturbance, labor strikes on a national level or labor strikes that impact the Rail Projects, freight embargoes and similar unavoidable circumstances beyond NICTD's or its agent's control and not caused by any of the following: (i) the negligent or intentional act or omission of NICTD, its agents, contractors or consultants (at whatever tier), or anyone else for whom NICTD is responsible, (ii) breach of this Governance Agreement by NICTD, its agents, contractors or consultants (at whatever tier), or anyone else for whom NICTD, its agents, contractors or consultants (at whatever tier), or anyone else for whom NICTD is responsible, contractors or consultants (at whatever tier), or anyone else for whom NICTD is responsible.
  - 1.47 **FTA** means the Federal Transit Administration.
- 1.48 **FTA CIG Program** means the Federal Transit Administration Capital Investment Grants Program.

- 1.49 **FTA Financial Plans** means the financial plans prepared by NICTD as required by the FTA CIG Program and that are approved in writing by IFA and RDA.
- 1.50 **GANS** means the grant anticipation notes issued by IFA pursuant to IC 5-1.3-6 for the purpose of financing a portion of the Project Costs, which GANS are to be paid from reimbursements received from FTA under the FFGA or as otherwise provided in IC 8-14-15.2.
  - 1.51 **Grants** has the meaning set forth in Section 6.1(b).
  - 1.52 **Gross Revenues** has the meaning set forth in <u>Section 6.1(a)</u>.
  - 1.53 **IFA** means the Indiana Finance Authority.
- 1.54 **IFA Bonds** means the bonds, notes or other obligations, including Draw Down Bonds and GANS, issued by IFA for the Rail Projects described in <u>Section 5.2</u>, including bonds, notes or other obligations issued to refund or refinance such bonds, notes or other obligations.
- 1.55 **IFA Initial Notice** means IFA's notice of a NICTD Operations Adverse Condition pursuant to Section 11.5.
- 1.57 **IFA-RDA GANS Master Lease (West Lake Project)** means the GANS Master Lease Agreement (West Lake Project), which is expected to be entered into between IFA, as lessor, and RDA, as lessee, in connection with the issuance of the GANS.
- 1.58 **IFA-RDA Master Lease** means with respect to the West Lake Corridor Project, the IFA-RDA Available Revenue Master Lease (West Lake Project), or the IFA-RDA GANS Master Lease (West Lake Project), and as either such Master Lease may be further amended or supplemented from time to time, and with respect to the Double Track Project, the IFA-RDA Master Lease (Double Track Project) or any Master Lease substantially in the form of the IFA-RDA Master Lease (Double Track Project), and as such Master Lease may be further amended or supplemented from time to time.
- 1.59 **IFA-RDA Master Lease (Double Track Project)** means the RDA Member Dues Master Lease Agreement (Double Track Project), dated as of March 1, 2020, as amended and supplemented by the Supplement to RDA Member Dues Master Lease Agreement (Double Track Project), dated as of March 1, 2021, between IFA, as lessor, and RDA, as lessee.
  - 1.60 **Indiana Design-Build Statute** means IC §5-30-1 et seq.
- 1.61 **Indemnitors** means the following entities who undertake a duty to indemnify a Party hereunder: (i) nonparty consultants, contractors, designers, design-builders, and other professionals involved in the design and/or construction of the Rail Projects, as provided in <u>Section 7.2.6.1</u>; and (ii) NICTD, as provided in <u>Section 15.6</u>.

- 1.62 **INDOT** means the Indiana Department of Transportation.
- 1.63 **Interlocal Agreement** shall mean one or more agreements entered into by approximately 16 governmental entities in Lake County in which each governmental entity agrees to make payments to support the costs of the West Lake Corridor Project, some or all of which payments shall be directed to the RDA as provided in IC 6-3.6-9-5, IC 6-3.6-11-5.5, IC 6-3.6-11-7. IC 36-7.5-4-2.5 and IC 36-7.5-4-20.
- 1.64 **Leased Premises** means that portion of the improvements constructed in connection with the Rail Projects and the real property upon which such improvements shall be constructed, together with the existing rail lines and any real and personal property acquired after the execution of Leases, all as more fully described in the Leases, and shall not include any existing stations, existing parking facilities, existing buildings, or NICTD rolling stock, except to the extent costs related thereto are included in the applicable Cost Budget.
- 1.65 **Leases** means, collectively, the NICTD-RDA Ground Leases, the RDA-IFA Underlying Leases, the IFA-RDA Master Leases and the RDA-NICTD Operating Leases.
  - 1.66 **Level of Service** has the meaning set forth in Section 8.7.
- 1.67 **Major Contract** means a contract pertaining to the Rail Projects and that is (i) in excess of \$500,000, or (ii) a contract with an entity, when contracts with that entity or any related person or affiliate thereof in aggregate exceed \$500,000, or (iii) any of a series of contracts entered into by NICTD for the purpose of avoiding the determination of whether the contracts would be a Major Contract pursuant to clause (i) or (ii) hereof.
  - 1.68 **Maximum Debt Service** has the meaning set forth in Section 6.3.
- 1.69 **Maximum NICTD Excess Cost Contribution** means an amount of up to \$45 Million, which amount shall be separate from and in addition to the NICTD West Lake Delay Cost Contribution, the NICTD Double Track Cost Contribution and the NICTD Double Track Delay Cost Contribution.
- 1.70 **Member Dues** means the portion of the annual payments to RDA made by the Cities of Gary, East Chicago and Hammond and the Counties of Lake and Porter pursuant to IC 6-3.6-6-9, IC 6-3.6-11-6(b), IC 36-7.5-3-1.7 and IC 36-7.5-4-2, to the extent required to be deposited in the fund established under IC 36-7.5-4-1 pursuant to the Revenue Trust Fund Agreement.
- 1.71 **METRA Track 4 Project** means the construction of improvements proposed and generally described as a new and/or rehabilitated main line Track 4 in Chicago, Illinois, between Eleventh (11<sup>th</sup>) Place and Millennium Station, a high level platform at Van Buren Street Station to serve Track 4 including extending the Van Buren Street Station tunnel, with elevator access to the Track 4 platform to meet Americans with Disabilities Act requirements, with stairs to Jackson Boulevard from the new platform, replacement mid-day storage for Metra and NICTD trains, construction of a new high level platform along Track 14 at Millennium Station for normally exclusive use by NICTD trains, and such other track, signal, platform, and catenary improvements allowing for improved ingress and egress to the upper level of Millennium Station, all of which

will enhance NICTD's service in and out of Chicago after completion of the Double Track Project and the West Lake Corridor Project.

- 1.72 **NEPA** means National Environmental Policy Act, 42 U.S.C. § 4321 et seq.
- 1.73 **NICTD** means the Northern Indiana Commuter Transportation District.
- 1.74 **NICTD Double Track Cost Contribution** means an amount of up to \$10,681,502 that NICTD agrees to make available to cover Project Costs for the Double Track Project up to the amount of the final \$10,681,502 of the Double Track Project Cost Budget as of the Effective Date as set forth in the Financing Plan (Exhibit 5A) for the Double Track Project. The NICTD Double Track Cost Contribution will be used at the IFA's discretion for the Double Track Project and shall be provided only after all other funding sources provided for in the Financing Plan for Project Costs of the Double Track Project within the Double Track Project Cost Budget have been exhausted. The NICTD Double Track Cost Contribution shall not be applied to Excess Construction Costs. The NICTD Double Track Cost Contribution is separate from the NICTD Excess Cost Contribution, the NICTD West Lake Delay Cost Contribution and the NICTD Double Track Delay Cost Contribution.
- 1.75 **NICTD Double Track Delay Cost Contribution** means NICTD's agreement to provide additional funding sufficient to pay for Double Track Delay Costs in accordance with Section 7.8.4(i) hereof, which funding shall be separate from and in addition to the Maximum NICTD Excess Cost Contribution. The NICTD Double Track Delay Cost Contribution may be provided through the issuance of bonds by NICTD, through the provision of cash on hand of NICTD, or through some combination thereof.
- 1.76 **NICTD Excess Cost Contribution** means NICTD's agreed contribution for Excess Construction Costs pursuant to Section 7.8, which amount shall be separate from and in addition to the NICTD West Lake Delay Cost Contribution, the NICTD Double Track Cost Contribution and the NICTD Double Track Delay Cost Contribution.
- 1.77 **NICTD Financial Operations Report** means the annual report to be prepared by NICTD and provided to IFA and RDA pursuant to <u>Sections 6.1 and 6.3.</u>
- 1.78 **NICTD Initial Notice** means NICTD's notice of a NICTD Operations Adverse Condition under <u>Section 8.9</u>.
- 1.79 **NICTD Operations Adverse Condition** means the occurrence of any of the conditions listed in Section 11.5.
- 1.80 **NICTD-RDA Ground Lease** means, with respect to the West Lake Corridor Project, the Ground Lease Agreement (West Lake Project), dated as of February 1, 2020, as amended and supplemented by the Supplemental Ground Lease Agreement (West Lake Project), dated as of July 1, 2020, and the Second Supplemental Ground Lease Agreement (West Lake Project), dated as of July 1, 2021, each between NICTD, as lessor, and RDA, as lessee, and as such Ground Lease may be further amended or supplemented from time to time, and with respect to the Double Track Project, the Ground Lease Agreement (Double Track Project), dated as of March 1, 2020, as amended and supplemented by the Supplemental Ground Lease Agreement (Double

Track Project), dated as of March 1, 2021, both between NICTD, as lessor, and RDA, as lessee, and as such Ground Lease may be further amended or supplemented from time to time.

- 1.81 **NICTD West Lake Delay Cost Contribution** means NICTD's agreement to provide additional funding sufficient to pay for WL Delay Costs in accordance with Section 7.8.3(i) hereof, which funding shall be separate from and in addition to, and over and above, the Maximum NICTD Excess Cost Contribution. The NICTD West Lake Delay Cost Contribution may be provided through the issuance of bonds by NICTD, through the provision of cash on hand of NICTD, or through some combination thereof.
  - 1.82 **O&M** means operation and maintenance.
- 1.83 **Operations Cure Period** means the period provided to NICTD following approval by IFA and RDA of the Recovery Plan in which to cure a NICTD Operations Adverse Condition as provided in <u>Section 11.6.4</u>.
  - 1.84 **Operations Expenses** has the meaning set forth in Section 6.1(d).
  - 1.85 **Operations Step-In Notice** means the notice by IFA provided for in <u>Section 11.9</u>.
- 1.86 **Operations Step-In Period** means the period from the date of Operations Step-In Notice through Termination.
- 1.87 **Operations Step-In Rights** means the rights of IFA to step in and assume operation and maintenance of the South Shore Line pursuant to <u>Section 11.9.</u>
- 1.88 **Participant Unit Revenues** means: (a) amounts withheld by the auditor of State from certain municipalities within Lake County, as determined by the Indiana Department of Local Government Finance, and paid to the Secretary-Treasurer of RDA pursuant to IC 6-3.6-11-5.5; and (b) revenues derived from the Interlocal Agreement or from other actions taken by certain municipalities within Lake County offering to provide revenue to support and finance the West Lake Corridor Project pursuant to IC 6-3.6-11-7.
- 1.89 **Participating Entities** means the governmental entities located in Lake County, Indiana who are parties to the Interlocal Agreement, including any such entities who hereinafter agree to participate therein.
- 1.90 **Payment Request** means a NICTD request for payment or reimbursement of Project Costs incurred for the Rail Projects.
  - 1.91 **Permitted NICTD Debt** has the meaning set forth in Section 6.3.
- 1.92 **Post-Acceptance Disbursement Date** means, with respect to the applicable Rail Project, the later of: (a) the date six months after the date of Project Closeout; or (b) the first date upon which: (i) all contractor claims have been resolved with respect to the Rail Project; and (ii) all required FTA deliverables with respect to the Rail Project have been satisfied.

- 1.93 **Project Closeout** means with respect to the Double Track Project the date upon which the FFGA for the Double Track Project is closed, final payment under the FFGA for the Double Track Project has been made, the final report prepared in accordance with the before-and-after study plan for the Double Track Project (as described in the FFGA for the Double Track Project) has been delivered to FTA, the scope described on Exhibit 1 attached hereto has been completed and all work under DT 1, DT 2, DT 3, DT 4, DT 5, DT 6, and DT 7, including in each instance all punch list items have been completed, and with respect to the West Lake Corridor Project the date upon which the FFGA for the West Lake Corridor Project is closed, final payment under the FFGA for the West Lake Corridor Project has been made, the final report prepared in accordance with the before-and-after study plan for the West Lake Corridor Project (as described in the FFGA for the West Lake Corridor Project) has been delivered to FTA, the scope described on Exhibit 2 attached hereto has been completed and all work under the WL Design-Build Contract, including all punch list items have been completed. The end date for Project Closeout for the Double Track Project is set forth on Exhibit 4A attached hereto. The end date for Project Closeout for the West Lake Corridor Project is set forth on Exhibit 4B attached hereto.
- 1.94 **Project Contingency** means the dollar amount(s) set forth in the applicable Cost Budget, over and above the estimated Project Costs, as of the Effective Date, whether allocated to specific aspects of the Rail Projects or unallocated, which have been or may be used for Project Costs as permitted by this Governance Agreement.
- 1.95 **Project Costs** means all direct and indirect actual costs for completion of design, construction and development of each of the Rail Projects. Project Costs does not include such costs related to the "Alternates" as defined in the WL Design-Build Contract, unless specifically approved in writing by IFA and RDA.

#### 1.96 **Project Executives** means:

- (a) The CEO of the RDA, or delegate;
- (b) The President and CEO of NICTD, or delegate; and
- (c) The Public Finance Director of the State of Indiana (in the capacity as the executive director of IFA), or delegate.
- 1.97 **Project Schedule** means the agreed time schedule for completion of development, design and construction of each of the Rail Projects, attached as <a href="Exhibit 4A">Exhibit 4A</a> (Double Track Project Schedule) and <a href="Exhibit 4B">Exhibit 4B</a> (West Lake Corridor Project Schedule), as applicable, which includes dates for major milestones, including for the start of Revenue Services Operations and Substantial Completion, and reasonable float times for unanticipated contingencies. Each Project Schedule may be updated from time to time during construction of the Rail Projects as the Parties may agree in writing.

#### 1.98 **Rail Projects** means

- (a) the Double Track Project and
- (b) the West Lake Corridor Project.

- 1.99 **Rail Project** means either the Double Track Project or the West Lake Corridor Project, depending on the context.
- 1.100 **Rail Projects Improvements** means the improvements to be constructed in connection with the Rail Projects.
- 1.101 **Rail Projects Property** means all real and personal property interests, including but not limited to Leased Premises, Rail Projects Improvements, and such easements, licenses, leases, or property rights, in whatever form reasonably necessary to construct the Rail Projects and operate the South Shore Line, but not including existing stations, existing parking facilities, or existing buildings, or NICTD rolling stock, except to the extent any costs for rolling stock are included in the applicable Cost Budget).
  - 1.102 **RDA** means the Northwest Indiana Regional Development Authority.
- 1.103 **RDA Excess Cost Contribution** means an amount not to exceed \$30 Million that RDA agrees to make available to the extent necessary to cover costs for the West Lake Corridor Project that are in excess of the Cost Budget as of the Effective Date, to be used only to the extent NICTD is current on its obligations to pay ongoing WL Delay Costs as they are incurred to the extent the West Lake Corridor Project Cost Budget has been exceeded as required by Section 7.8.3(i)(a) and after NICTD has paid all of its obligations required in Section 7.8.3(i)(b) and Section 7.8.3(ii). The RDA Excess Cost Contribution shall not be used to pay WL Delay Costs.
- 1.104 **RDA-IFA Underlying Lease** means, with respect to the West Lake Corridor Project, the Underlying Lease Agreement (West Lake Project), dated as of February 1, 2020, as amended and supplemented by the Supplemental Underlying Lease Agreement (West Lake Project), dated as of July 1, 2020, and the Second Supplemental Underlying Lease Agreement (West Lake Project), dated as of July 1, 2021, each between RDA, as lessor, and IFA, as lessee, and as such Underlying Lease may be further amended or supplemented from time to time, and with respect to the Double Track Project, the Underlying Lease Agreement (Double Track Project), dated as of March 1, 2020, as amended and supplemented by the Supplemental Underlying Lease Agreement (Double Track Project), dated as of March 1, 2021, both between RDA and IFA, and as such Underlying Lease may be further amended or supplemented from time to time.
- 1.105 **RDA/IFA Project Costs** means RDA's and IFA's internal costs, consultant costs and financing costs for the Rail Projects.
- 1.106 **RDA Initial Notice** means RDA's notice of a NICTD Operations Adverse Condition under <u>Section 11.5</u>.
- 1.107 **RDA-NICTD Operating Lease** means, with respect to the West Lake Corridor Project, the Operating Lease Agreement (West Lake Project), dated as of February 1, 2020, as amended and supplemented by the Supplemental Operating Lease Agreement (West Lake Project), dated as of July 1, 2020, and the Second Supplemental Operating Lease Agreement (West Lake Project), dated as of July 1, 2021, each between RDA, as lessor, and NICTD, as lessee, and as such Operating Lease may be further amended or supplemented from time to time, and with respect to the Double Track Project, the Operating Lease Agreement (Double Track Project), dated as of March 1, 2020, as amended and supplemented by the Supplemental Operating Lease Agreement

(Double Track Project), dated as of March 1, 2021, both between RDA, as lessor, and NICTD, as lessee, and as such Operating Lease may be further amended or supplemented from time to time.

- 1.108 **Recovery Plan** means NICTD's written recovery plan setting forth NICTD's plans and proposed steps to cure NICTD Operations Adverse Condition(s).
  - 1.109 **Restricted Taxes** has the meaning set forth in <u>Section 6.1</u>.
- 1.110 **Revenue Coverage** means the calculation of the revenue coverage ratio as provided in <u>Section 6.3</u> and <u>Exhibit 7</u> and certified by NICTD in a manner consistent with <u>Exhibit 7</u>, and subject to verification by the IFA and/or RDA in their discretion as provided in <u>Section 6.8</u>.
- 1.111 **Revenue Service Operations Date** means the date by when NICTD shall commence revenue operations for the West Lake Corridor Project or the Double Track Project, as applicable, which date is set forth in the applicable Project Schedule.
- 1.112 **Revenue Trust Fund** means the trust fund established and maintained to receive, hold and disburse Rail Projects funds, including FTA grant funds, bond proceeds, Member Dues, Participant Unit Revenues, State appropriations and local contributions according to the terms of this Governance Agreement.
- - 1.114 **ROW** means right of way.
- 1.115 **SSI Policy** means the terms and conditions of the NICTD Policy and Procedures on Sensitive Security Information as adopted by NICTD on January 26, 2018, a copy of which is attached as Exhibit 9.
- 1.116 **South Shore Line** means the existing South Shore Line Passenger Railroad, a commuter train service between South Bend, Indiana and Millennium Station in Chicago, Illinois, including: (1) those portions of the Double Track Project when complete and those portions of the West Lake Corridor Project when complete; (2) those assets being funded under the FTA CIG Program; and (3) all other assets, whether now owned or hereafter acquired or leased by NICTD for operation of the South Shore Line.
  - 1.117 **State** means the State of Indiana.
- 1.118 **State Appropriation Funds** means funds appropriated to the IFA for State fiscal year 2021 by the Indiana General Assembly pursuant to Public Law 165-2021 for the Rail Projects.
- 1.119 **State Budget Director** means the State Budget Director of the Indiana State Budget Agency or in the absence of this position, the person designated by the Governor to serve in the capacity of the State Budget Director under this Governance Agreement.

- 1.120 **Subsidy Shortfall** means the amount of any insufficiency of the Escrow Fund to cover any West Lake Corridor Project Deficit in any quarter pursuant to <u>Section 8.4.4</u>.
- 1.121 **Substantial Completion** means the date by when the Design-Builder or the Double Track Contractors shall achieve substantial completion of the West Lake Corridor Project or the Double Track Project, respectively, which date is set forth in the applicable Project Schedule.
  - 1.122 **Taxes** has the meaning set forth in <u>Section 6.3</u>.
- 1.123 **Tax Credit Dollars** means the funds received by RDA pursuant to IC 6-3.1-20-7(d) from the Indiana Department of Revenue as described in Section 8.4.1.
  - 1.124 **TDD** means Transit Development District/s.
- 1.125 **Termination** means the termination or expiration of this Governance Agreement as provided in <u>Article 13</u>.
  - 1.126 **Unrestricted Taxes** has the meaning set forth in Section 6.1(a).
- 1.127 **West Lake Corridor Project** means an approximately 8-mile new north/south passenger rail line connected to the existing South Shore Line, with four stations, between Hammond (Gateway Station) and Munster/Dyer (Main Street Station, including all property owned or leased by NICTD for the operation of the South Shore Line over the West Lake Corridor Project), all as more fully described in Exhibit 2.
- 1.128 **West Lake Deficit Cost Submission** means the NICTD request for payment for West Lake Corridor Operating Deficits, if any, as provided in <u>Section 8.4.3</u> and in compliance with the West Lake Corridor Deficit Cost Submission Form attached hereto as <u>Exhibit 8</u>, accompanied by a completed form of <u>Exhibit 6</u> that is certified by NICTD and subject to verification by the IFA and/or RDA in their discretion.
- 1.129 **West Lake Operating Deficit** means the operating cost deficit, if any, related to the West Lake Corridor Project (direct and indirect costs of NICTD, including administrative overhead and costs associated with operating and maintaining the West Lake Corridor Project after completion, minus all revenue associated or derived from the West Lake Corridor Project) as calculated as set forth in Exhibit 6, which shall be subject to verification by IFA and/or RDA in their discretion.
- 1.130 **West Lake Portion of the Maximum NICTD Excess Cost Contribution** means not more than \$30,000,000 of the Maximum NICTD Excess Cost Contribution, which is allocated to the West Lake Corridor Project as provided in <u>Section 7.8.2</u>.
- 1.131 **West Lake Portion of the NICTD Excess Cost Contribution** means the portion of the NICTD Excess Cost Contribution allocable to the West Lake Corridor Project as provided in Section 7.8.2.
- 1.132 **WL Deficit Alternative Funding** means funding secured from the General Assembly in an amount that is equal to or exceeds that year's West Lake Operating Deficit,

whether received by or provided to any Party or any other entity, in each case in any manner that is dedicated or made available for use to subsidize the West Lake Operating Deficit.

- 1.133 **WL Delay Costs** means any of the following costs incurred as a result of delay in construction of the West Lake Corridor Project: (i) damages or costs incurred as a result of a delay in Substantial Completion that are in excess of the liquidated damages that the Design-Builder is liable for pursuant to Section 9.7 of the WL Design-Build Contract; and/or (ii) damages or costs incurred pursuant to Section 12.9 of the WL Design-Build Contract.
- 1.134 **WL Design-Build Contract** means the Design-Build Contract For Design-Build Services, dated October 29, 2020, between NICTD and the Design-Builder, for the West Lake Corridor Project.

#### **ARTICLE II.**

## **CONDITIONS PRECEDENT**

- 2.1 The following are conditions precedent to this Governance Agreement being effective as to each Rail Project:
  - 2.1.1 Agreement to the Cost Budgets, attached as Exhibits 3A and 3B;
  - 2.1.2 Agreement to the Project Schedules, attached as Exhibits 4A and 4B;
  - 2.1.3 Agreement to the Financing Plans, attached as Exhibits 5A and 5B; and
  - 2.1.4 Approval of this Governance Agreement, including all exhibits hereto, by the boards or authorized representatives of each of the Parties.

#### **ARTICLE III.**

#### PROJECT MANAGEMENT AND COORDINATION

- 3.1 The Rail Projects shall be managed by the Project Executives in the manner herein provided, who shall conduct bi-weekly progress meetings to report on the progress of the Parties' respective duties and obligations as set forth in this Governance Agreement and any Development Agreement that may be entered into.
  - 3.1.1 Where and when possible, decisions on the Rail Projects shall be determined by agreement among the Project Executives. Where such agreement is not able to be achieved, the dispute resolution procedures in Article 10 shall apply.
  - 3.1.2 In making decisions regarding the Rail Projects, the Project Executives shall consider factors that include: (1) cost containment; (2) meeting FTA grant criteria and satisfying grant assurances; (3) the applicable Project Schedule, including the timeliness of decision making during the funding, design and construction phases; (4) incremental revenue opportunities; (5) market acceptance; (6) the Parties' contractual obligations and duties to third parties, including their respective contractual risks, duties and liabilities; (7)

the impact on current and future FTA grants on NICTD, RDA and IFA projects; (8) impact on future TDD; (9) the State's bond rating; and (10) the availability of funds. These factors are collectively referred to herein as "Decision Making Framework Factors".

- 3.2 NICTD and RDA shall consult with IFA in advance of taking any action or making any decision, regarding the Rail Projects that would, in the judgment of IFA, materially impact a bond issuance or the security for a bond issuance, in which IFA acts as the issuer, and such action and/or decision shall be subject to IFA approval.
- 3.3 NICTD, RDA and IFA shall participate in bi-weekly progress meetings where NICTD shall update RDA and the IFA on the status and progress of the Rail Projects, including prompt updates as to any developments of significance that may impact the applicable Cost Budget or Project Schedule. NICTD, RDA and IFA may request more frequent updates at their individual discretion.
- 3.4 Prior to the issuance of any bonds or other financing of the Rail Projects, the Parties may enter into a Development Agreement, which, among other things may address details regarding the applicable Financing Plan.

#### 3.5 Federal Coordination

3.5.1 RDA, IFA and NICTD shall work together to coordinate overall federal strategy, interface, and implementation.

#### 3.5.2 Federal Transit Administration

- 3.5.2.1 NICTD, with assistance of RDA, shall be the lead agency in the FTA grant application preparation process and interfacing with the FTA, except as otherwise provided herein.
- 3.5.2.2 NICTD, RDA and IFA representatives involved with or assigned to the Rail Projects shall work together to assure that NICTD, RDA and IFA are timely informed of any regularly scheduled meetings, submittals, or calls with FTA to help secure full involvement, awareness and cooperation regarding interactions with FTA, including but not limited to the FTA grant application process. NICTD, RDA and IFA, and their representatives, shall jointly coordinate and participate on all FTA regularly scheduled meetings, conference calls and submittals. The Parties also shall promptly inform each other as to any potential issues or concerns relating to, or raised by, FTA, and shall inform each other as to all substantive communications with FTA.
- 3.5.2.3 NICTD, RDA and IFA shall share with each other and their respective representatives, copies of all draft documents to be provided to FTA, with adequate opportunity (no less than 15 calendar days, unless otherwise agreed in writing) to review in advance of submission to FTA, and also shall promptly provide copies of all documents exchanged with FTA.

### 3.5.2.4 FTA Grant Application

- 3.5.2.4.1 Subject to the provisions herein, NICTD shall be the applicant of record on the FTA grant program applications regarding the Rail Projects and shall lead such grant application process.
- 3.5.2.4.2 NICTD, RDA and IFA shall work together to identify potential issues related to the FTA grant application process, shall keep each other informed of progress on the FTA grant applications, and shall promptly inform each other of substantive communications with FTA regarding the FTA grant applications. NICTD, RDA and IFA commit to each other to promptly and fully participate in the FTA grant application process.
- 3.5.2.4.3 RDA and/or IFA may hire technical consultant(s) to conduct independent review(s) of grant application materials and support the grant application process. NICTD shall cooperate with any such technical consultant hired by RDA and/or IFA. IFA and RDA shall have the right to review, comment and approve any grant application materials prior to submission to FTA.
- 3.5.2.4.4 To the maximum extent possible, all costs incurred by RDA and IFA, including but not limited to financing costs, interest, costs for professional services and legal fees as estimated, identified and included in the applicable Cost Budget or necessarily incurred with respect to the Rail Projects, shall be reimbursable with FTA grant funds, other funds (not including funds from NICTD other than as set forth in this Governance Agreement), or bond proceeds. RDA, IFA and NICTD shall timely exchange information with each other as to the types and amounts of all estimated future Project Costs so that they are sufficiently identified and can be included in the cost estimate submitted to and approved by FTA. In order to help maintain eligibility for cost reimbursement under the FTA grants, all Parties shall report to each other monthly as to the costs that are being incurred and the basis for asserting such costs are eligible for reimbursement with an FTA grant. Without regard to whether any costs incurred by IFA or RDA are reimbursable by FTA grant funds, IFA and RDA may be reimbursed from the proceeds of the IFA Bonds or the rental payments due under the Leases in their sole and absolute discretion.

#### 3.6 State Coordination

- 3.6.1 RDA and NICTD shall coordinate with the IFA and the State executive and legislative branches, as necessary, to keep them informed of project progress.
- 3.6.2 NICTD shall interface with INDOT on technical matters, as necessary, and shall provide RDA updates and opportunity to comment with respect to such matters.

#### 3.7 Local Government Coordination

- 3.7.1 RDA shall coordinate with local units of government on local funding commitments. NICTD shall provide support.
- 3.7.2 NICTD is responsible for the NEPA process and, therefore, shall be responsible for coordinating alignment and technical decisions with local units and communities of the Rail Projects. NICTD shall advise RDA in advance, and RDA shall be provided an opportunity to comment in advance, with respect to such plans.
- 3.7.3 RDA, NICTD, and IFA agree they will jointly cooperate and proceed in good faith to attempt to secure any additional funds from the cash participants pursuant to and in accordance with I.C. 36-7.5-4.5-16 and to deposit such funds in the Revenue Trust Fund.

#### 3.8 Community Engagement

- 3.8.1 NICTD shall conduct local community outreach regarding the technical elements of the Rail Projects (e.g., station location, bike paths, station design, parking, etc.). NICTD's outreach shall be consistent and coordinated with RDA's development plans regarding TDDs.
- 3.8.2 NICTD shall coordinate with local units of government on ancillary infrastructure improvements to facilitate the Rail Projects.
- 3.8.3 NICTD shall update RDA and IFA regularly through the bi-weekly progress meetings on local government and community engagement.

#### 3.9 Private Business Use Agreements

- 3.9.1 The IFA has provided NICTD with advance consent to enter into Insubstantial Future Purported Private Business Use Agreements (as defined in such consent) regarding the Double Track Project. This consent allows NICTD to enter into agreements on and after this date, which purport to result in Private Business Use of the Facilities and Lease Premises, so long as: (1) pursuant to the terms of any such agreement, it will not generate revenues for, or result in payments to, NICTD in the amount of \$100,000 or more in any one year; and (2) any such agreement will not constitute or result in an Impermissible Encumbrance (as described in Section 11(a) of the Ground Lease Agreement (Double Track Project)). This advance consent may be revoked at any time by the IFA.
- 3.9.2 The IFA hereby provides its advance consent for NICTD to enter into insubstantial future purported private business use agreements regarding the West Lake Corridor Project, which purport to result in Private Business Use of the Facilities and Lease Premises, subject to the following conditions: (1) pursuant to the terms of any such agreement, it will not generate revenues for, or result in payments to, NICTD in the amount of \$100,000 or more in any one year; and (2) any such agreement will not constitute or result in an Impermissible Encumbrance (as described in Section 11(a) of the Ground Lease Agreement (West Lake Project)). This advance consent may be revoked at any time by the IFA.

3.9.3 Prior to entering into any private business use agreement along the West Lake Corridor Project that are within one mile of any train station, NICTD shall coordinate with RDA and obtain RDA's written consent to such private business use agreement. RDA will review any such proposed private business use agreements to ensure maximum investment and alignment with TDD best practices.

Requests for consent to private business use agreements pursuant to this Section 3.9.3 shall be sent in writing by NICTD to RDA. Each such request shall include a substantially final draft of the proposed private business use agreement (e.g. the conveyance, lease, sublease, management or incentive payment contract, license or other agreement that may give rise to private business use), together with a description of the timing and the rationale for entering into such agreement (the "Proposed Private Business Use"). RDA shall conduct their review and approve or reject, in writing, such request within 30 days of receipt. If no response is made to a request for such consent to private business use within 30 days of receipt by RDA, then NICTD shall provide additional written notice to RDA that the Proposed Private Business Use will be deemed to be approved. If no response to this additional written notice is received within three business days of receipt of the additional written notice by RDA, then the request shall be deemed approved. In the event a request for consent to private business use pursuant to this Section 3.9.3 is deemed approved, NICTD shall have the right to proceed as though consent to the Proposed Private Business Use was expressly provided by RDA. In any event, upon the execution and delivery by NICTD of any such private business use agreement, NICTD shall provide RDA a copy of such executed agreement within 10 business days of the execution thereof.

#### ARTICLE IV.

#### PROPERTY INTERESTS AND ACTIVITIES

- 4.1 NICTD shall lead ROW acquisition activities for the Rail Projects. NICTD shall timely acquire or shall utilize its powers, once the requisite approvals for all aspects of real estate acquisition by RDA and IFA have been provided, including eminent domain and condemnation powers, to timely acquire the Rail Projects Property and shall timely add any such property to the properties leased to RDA under the applicable Lease.
  - 4.1.1 If agreed by the Parties, responsibility for ROW acquisition may be performed by a NICTD consultant or transferred to the successful design-builder or prime contractor, depending on final delivery and procurement structure.
  - 4.1.2 NICTD's ROW acquisition plans shall be subject to RDA's right to review and approve, with adequate time for review. RDA's review shall be to insure the Project scope, schedule and budget are met and to ensure consistency with TDD goals. Any RDA ROW modification requests must be timely made and any change requested in the ROW must be within the project scope, schedule and budget of the Rail Projects. Otherwise, RDA must secure additional funds to potentially cover any increased land acquisition costs and construction delay costs that may occur as a result of this RDA review process.

- 4.2 NICTD shall own the Rail Projects Property and shall make such Rail Projects Property available for development, construction, and operation of the Rail Projects through Termination, by including such Rail Projects Property in the property subject to the Leases.
  - 4.2.1 At such time as agreed to by the Parties, but in no event later than the issuance of any IFA Bonds,
    - (a) the Parties shall have entered into the Leases pursuant to which NICTD shall lease the Rail Projects Property to RDA and RDA shall lease the Rail Projects Property to IFA.
    - (b) IFA shall have entered into one or more IFA-RDA Master Leases, pursuant to which IFA shall lease the Rail Projects Property to RDA with respect to such IFA Bonds.
    - (c) RDA shall have entered into the RDA-NICTD Operating Leases, pursuant to which RDA shall lease the Rail Projects Property to NICTD, thus enabling NICTD to acquire, equip, construct, operate and maintain the Rail Projects.
  - 4.2.2 With respect to the Double Track Project, notwithstanding the IFA-RDA Master Lease (Double Track Project), IFA shall be authorized to make such changes thereto prior to the issuance of any additional IFA Bonds for the purpose of financing the costs of the Double Track Project, as deemed necessary or desirable by IFA and approved by RDA for the issuance of such IFA Bonds and the receipt of the most desirable amount of proceeds thereof.
  - 4.2.3 Upon repayment of all of the IFA Bonds, the Leases shall terminate or expire by their terms, except in the event that IFA has elected to exercise its Operations Step-In Rights in the manner provided in Section 11.9, in which case this Governance Agreement shall continue, unless and until one of the following occurs: (a) IFA agrees to no longer exercise its Operations Step-In Rights; or (b) upon the written request of NICTD to the State Budget Director, the State Budget Director determines, it its sole discretion, that the Leases may terminate or expire by their terms.
- 4.3 NICTD shall obtain all required federal and state permits for the Rail Projects and shall take the lead defending any challenges to such permits. RDA has the right, but not the obligation, to assist NICTD in defending any challenges to the permits and to cooperate with NICTD in that defense.
- 4.4 NICTD shall be responsible for all utility relocation processes for the Rail Projects. RDA has the right, but not the obligation, to participate in utility relocation.
  - 4.4.1 RDA has approval rights of final utility locations with respect to those utility relocations that impact future TDD opportunities. If RDA requires an alternative utility relocation based on its impact to future TDD opportunities, then any additional costs of such utility relocation are to be funded by RDA so that there are otherwise no cost increases or time delays to the applicable Cost Budget and Project Schedule.

- 4.5 NICTD shall be responsible for achieving and maintaining compliance with NEPA and with any environmental approval processes for the Rail Projects, including, but not limited to, any applicable Record of Decision, FONSI, or federal, state or local laws. RDA has the right, but not the obligation, to participate in the environmental approval process and compliance therewith.
  - 4.5.1 NICTD shall promptly: (a) inform RDA as to the status of all environmental issues and invite RDA to all meetings and conference calls with EPA, IDEM, the U.S. Army Corps of Engineers and/or other governmental entities; (b) provide RDA with draft copies of all documents to be provided to such entities in advance; and (c) provide RDA with copies of all documents exchanged with such entities.
- 4.6 NICTD shall be responsible for securing all necessary trackage rights for the Rail Projects.
- 4.7 NICTD shall be responsible for complying with all applicable state and federal requirements, including, but not limited to: FTA requirements, FFGA requirements, and FTA grant requirements, which include all FTA requirements as to reporting and/or providing other information.

# ARTICLE V.

#### FINANCING AND FUNDING

- 5.1 Subject to IC 5-1.3, RDA is the exclusive fiscal officer for the Rail Projects. RDA shall coordinate with IFA on funding and financing from State and local jurisdictions.
- 5.2 IFA shall act as issuer for any bond financing pursuant to IC 5-1.3-6 and IC 5-13-10.5-20. IFA shall be responsible to determine financings, including determining the timing and manner of issuance of the IFA Bonds, and disbursement of financing proceeds to RDA, NICTD or their assignees, as applicable, subject to market conditions. Except for the Draw Down Bonds, the IFA Bonds are not expected to be issued to provide financing for a Rail Project until an FFGA has been executed for that Rail Project. In connection with the issuance of the IFA Bonds, the RDA and NICTD shall cooperate in any manner with the issuance of the IFA Bonds, the compliance by the Parties with the provisions of the financing agreements entered into in connection with the IFA Bonds, including the Leases, and any continuing disclosure obligations and business interruption insurance required from the Parties in connection with the IFA Bonds. The IFA Bonds shall be payable from lease rentals payable under the IFA-RDA Master Leases to the extent that funds have been appropriated or are available to RDA in the manner provided in the IFA-RDA Master Leases and the applicable Financing Plan.
- 5.3 RDA and IFA have developed the Financing Plans, attached hereto as Exhibit 5A and 5B. Notwithstanding any other provisions herein, IFA's and RDA's obligations to provide funding and financing for the Rail Projects and for the repayment of IFA Bonds shall be limited to no more than the amounts set forth in the applicable Financing Plan and as set forth in the applicable Cost Budget as of the Effective Date, which may be amended only by written agreement of the Parties. The amount of proceeds derived from the IFA Bonds from the funding described in the applicable Financing Plan shall be determined at the time of issuance of the IFA Bonds,

depending on, among other matters, market conditions. As stated in the applicable Financing Plan, except with respect to the Draw Down Bonds, the IFA does not anticipate issuing bonds until receipt of FFGA Approval. All funds provided by RDA and/or IFA to NICTD prior to FFGA Approval shall be limited to payment for Project Costs that are included in the applicable Cost Budget (as amended and updated by written agreement of the Parties), and that also are individually and specifically approved in advance and in writing (both as to the timing and amount of such costs being incurred) by IFA and RDA. For clarity, notwithstanding any other provisions or approvals herein, prior to FFGA Approval, neither IFA nor RDA shall be obligated to provide funding for contracts or other costs that have not been individually and specifically approved in advance by IFA and RDA in writing.

- 5.4 RDA shall make rental payments semiannually from the Member Dues and the Participant Unit Revenues, as described in the applicable Financing Plan, subject to the provisions of the IFA/RDA Master Leases and the Revenue Trust Fund Agreement.
- 5.5 In the event the County of Lake or Porter or the City of East Chicago, Gary or Hammond do not transfer or do not cause to be transferred to RDA on or before July 20 of each year all the Member Dues with respect to the preceding State fiscal year required pursuant to IC 36-7.5-4-2(a), RDA will send a notice to such County or City on or before the following July 31 indicating that any such shortfall shall be paid by the following August 25, and if such shortfall is not paid, RDA will notify the Treasurer of State on or before the following September 1 to intercept such shortfall pursuant to IC 36-7.5-4-16(b).
- 5.6 The Parties shall establish a Revenue Trust Fund in which the Deposit Trustee shall receive, hold and disburse Rail Project funds, including FTA grant funds, the NICTD Excess Cost Contribution, the NICTD West Lake Delay Cost Contribution, the NICTD Double Track Cost Contribution, the NICTD Double Track Delay Cost Contribution, all State funding, Member Dues, Participant Unit Revenues, the RDA Excess Cost Contribution, ARPA Funds, the State Appropriation Funds, bond proceeds and local contributions, in the manner described in the Financing Plans. Any and all Rail Projects grant funds received from FTA or any other source shall be transferred to the Deposit Trustee to be held in the Revenue Trust Fund for use in support of the Rail Projects, provided that FTA requirements are maintained for the use and disposition of FTA grant funds.

#### 5.7 Disbursement of Funds

5.7.1 NICTD shall submit Payment Requests for Project Costs incurred in accordance with the applicable Cost Budget (along with such supporting material as IFA and/or RDA may reasonably require), to RDA and IFA for review and approval, by the tenth (10th) day of each month. Payment Requests shall be made not more frequently than monthly. IFA and RDA shall conduct their review concurrently and approve or reject Payment Requests within 14 days of receipt of a Payment Request. If no objection is made by either IFA or RDA (in writing) within 14 days, then the Payment Request shall be deemed approved after three business days' advance written notice to IFA and RDA by NICTD. Upon IFA and RDA approval, IFA shall submit NICTD's request for payment, along with RDA/IFA Project Costs, to the Deposit Trustee under the Revenue Trust Fund Agreement to issue payment in accordance with the procedures for disbursement of funds

as provided in the Revenue Trust Fund Agreement and the documents for the IFA Bonds and in accordance with the applicable Financing Plan. In the event that a payment request is deemed approved, NICTD shall retain the right to transmit the payment request to the Deposit Trustee should IFA not do so.

- 5.7.2 NICTD must adhere to all federal laws, regulations and procedures related to the administration and reimbursement of federal funding for the Rail Projects, including, but not limited to, submitting timely and complete requests for payment to the FTA and following all reporting procedures established by the FTA under the FFGA. If NICTD is required to be the recipient of federal funding under the FFGA, then NICTD shall deposit all federal funding received from the FTA for the Rail Projects into the appropriate accounts in accordance with the Revenue Trust Fund Agreement.
- 5.7.3 Disbursements for approved Payment Requests shall be made by the Deposit Trustee no later than 7 days after approval, unless withheld by the Deposit Trustee in accordance with the Revenue Trust Fund Agreement.
- 5.7.4 The contractual payment terms in the contracts entered into by NICTD for the Rail Projects shall take into account and not be inconsistent with the procedures and time requirements for RDA, IFA, and the Deposit Trustee to review and approve disbursement of funds for the Payment Requests (e.g., the provisions of this Section allow for a review, approval, and disbursement process). Contracts entered into by NICTD should take this into account so that penalties for payment delay are not incurred).
- 5.8 NICTD hereby assigns to RDA all rights, title and interest of NICTD in the Interlocal Agreement, including any revenues paid thereunder from the Participating Entities, and covenants and agrees that all rights, title and interest of NICTD in the Interlocal Agreement are hereby vested in RDA. NICTD further agrees to cooperate in any manner requested by RDA with respect to the administration of the Interlocal Agreement and the use of such revenues to support the financing contemplated from such revenues in the applicable Financing Plan, including seeking cooperation and support from Lake County, Indiana, with respect thereto.
- 5.9 The Parties agree that on the Post-Acceptance Disbursement Date with respect to the West Lake Corridor Project, any balance of the RDA Excess Cost Contribution remaining in the applicable account within the Revenue Trust Fund shall be released to RDA. Any other remaining unexpended balance in the Revenue Trust Fund for the West Lake Corridor Project or the Double Track Project, on the Post-Acceptance Disbursement Date with respect to the applicable Rail Project, shall be distributed as agreed by the Parties in writing, except as otherwise provided in this Governance Agreement, the Revenue Trust Fund Agreement or any Trust Indenture (as defined in the Revenue Trust Fund Agreement).

#### ARTICLE VI.

#### NICTD OPERATIONS COVENANTS

6.1 NICTD agrees to provide to RDA and IFA by April 30th of each year (commencing on April 30, 2019) a copy of NICTD's audited Comprehensive Annual Financial Report for the

preceding year, as well as a financial report ("NICTD Financial Operations Report") for such preceding calendar year of NICTD's financial performance for such calendar year, which includes itemization of:

- (a) all revenues, those taxes collected pursuant to the provisions of IC 8-3-1.5-20.5 and 20.6, except those defined as Restricted Taxes below ("Unrestricted Taxes"), and any other income received by or on behalf of NICTD, but excluding Restricted Taxes and Grants each as defined below (collectively "Gross Revenues");
- (b) federal and State capital grants received by or behalf of NICTD ("Grants");
- (c) Indefinite Situs Taxes and any other taxes that may not be used to pay Operations Expenses (defined below) ("Restricted Taxes");
- (d) NICTD operating and maintenance expenses, excluding noncash expenses such as depreciation and amortization ("Operations Expenses");
- (e) capital expenditures incurred by or on behalf of NICTD ("Capital Expenditures"); and
- (f) a list of all private business use agreements, into which NICTD has entered since the last annual report, and a copy of any such agreement(s) upon the request of IFA or RDA.
- 6.2 All amounts in <u>Section 6.1(a) through (e)</u> (collectively) above shall be annual amounts based on a calendar year (excluding any prior revenue balances).
- 6.3 Together with the NICTD Financial Operations Report, NICTD shall also provide a debt coverage ("Debt Coverage") calculation demonstrating the ratio of Restricted Taxes and Unrestricted Taxes ("Taxes") pledged to the debt service pursuant to the Bond Resolution and reported for such calendar year divided by the maximum annual debt service on outstanding NICTD debt in the then current or any future year ("Maximum Debt Service") and a revenue coverage ("Revenue Coverage") calculation demonstrating the ratio of Gross Revenues for such calendar year, together with current unrestricted cash, as of January 1 of such calendar year, to the sum of Operations Expenses for such calendar year, all as reported in NICTD's audited financial statements for its previous fiscal year. As used herein "Permitted NICTD Debt" shall mean: (i) existing parity NICTD debt; (ii) contemplated GARVEE bonds for the METRA Track 4 Project in the maximum par amount of not more than \$120 Million, the payment of the principal of and interest on which can be paid by a pledge by NICTD of a maximum amount of \$8 Million per year, 80% of which is payable from Section 5337 State of Good Repair grants and 20% of which may be payable from Taxes over the applicable bond term; (iii) parity senior debt permitted by IFA and RDA pursuant to the Bond Resolution, subject to Section 6.4; (iv) debt incurred in a manner consistent with the requirements of Section 6.4; (v) issuance of debt to fund the NICTD Excess Cost Contribution, the NICTD West Lake Delay Cost Contribution, the NICTD Double Track Cost Contribution, and the NICTD Double Track Delay Cost Contribution; and (vi) debt incurred to refinance or refund Permitted NICTD Debt in accordance with Section 6.4.

- or Taxes for any purpose (other than Permitted NICTD Debt) without the prior written consent of the RDA and IFA: (i) during the period until the construction and development of both Rail Projects are completed and are operational; or (ii) thereafter, unless NICTD can demonstrate Debt Coverage of at least 2.0, after taking into account the debt service of such additional debt and all other debt RDA and IFA expect NICTD will need to issue to fund Capital Expenditures contemplated in NICTD's most recent five-year capital plan submitted to FTA. Notwithstanding the foregoing, NICTD may refund Permitted NICTD Debt, at any time, without IFA and RDA consent provided that: (i) the final maturity of the refunding debt does not exceed the final maturity of the Permitted NICTD Debt to be refunded, and (ii) the average annual debt service on the refunding debt does not exceed the average annual debt service on the Permitted NICTD Debt to be refunded.
- 6.5 NICTD shall manage its operations in each calendar year so that NICTD has a Revenue Coverage in such calendar year of not less than the 1.12 required for the FTA grant application.
- 6.6 Revenue Coverage and the Debt Coverage shall be calculated according to the form of calculation attached as Exhibit 7.
- 6.7 Except as otherwise provided in <u>Article 6</u>, NICTD may use its Gross Revenues, Restricted Taxes and Grants in any manner determined by NICTD in its discretion.
- 6.8 NICTD's Financial Operations Report, Debt Coverage calculation and Revenue Coverage calculation shall be subject to verification by the IFA and/or the RDA in their discretion.

#### **ARTICLE VII.**

#### DESIGN, CONSTRUCTION AND DEVELOPMENT OF THE RAIL PROJECTS

- 7.1 NICTD shall complete and deliver the Rail Projects in accordance with the applicable Cost Budget (as amended by written agreement of the Parties) and applicable Project Schedule (as amended by written agreement of the Parties), and in compliance with the limitations in this Governance Agreement concerning the use of Project Contingency.
- 7.2 NICTD, with the assistance of RDA, shall be the procuring entity for the design, acquisition, construction, construction management/owner's representative, and/or program management of the Rail Projects.
  - 7.2.1 NICTD shall make a recommendation to RDA and IFA regarding procurement and delivery structure/s for each Rail Project. Such recommendation shall be subject to RDA and IFA review and approval. IFA shall provide support to RDA on financing and commercial elements, as necessary, subject to the applicable Financing Plan.
  - 7.2.2 NICTD shall lead the review of design plans, technical proposals, reports, and related technical documents as necessary or helpful to facilitate the procurements. Subject to the SSI Policy, RDA shall have the right, but not the obligation, to participate.

- 7.2.3 The Parties acknowledge and agree that representatives of RDA and IFA, and their respective designees, who are exercising rights and functions described or contemplated in this Governance Agreement are persons with a "need to know" under the SSI Policy and are entitled to receive and review sensitive security information, provided they comply with the requirements of the SSI Policy. RDA and IFA will provide to NICTD (updated as necessary) a list of representatives and designees that they identify as "need to know".
- 7.2.4 All procurement proposal or bid evaluations related to the design, construction, construction management/owner's representative, and/or program management of the Rail Projects that are being procured through the design-build project delivery process shall involve a coordinated team of evaluators selected by NICTD that shall constitute the Technical Review Committee in conformance with the Indiana Design-Build Statute. RDA and IFA each shall be entitled to representation on the Technical Review Committee (a majority of the membership of which will be appointed by NICTD) and/or any other committee or body formed for purposes of evaluating the proposals or bids at any stage of the procurement process.
  - (a) NICTD shall lead the technical proposal evaluations and conduct the selection process through solicitation of statements of qualifications and qualitative and price proposals in accordance with the Indiana Design-Build Statute and the SSI Policy. RDA and IFA shall have the right, but not the obligation, to attend evaluation sessions and interviews, subject to the provisions in the SSI Policy.
  - (b) For procurements utilizing the design-bid-build competitive bidding project delivery system, bids and proposals shall be evaluated by NICTD in accordance with the statutory requirements. RDA and IFA shall have the right, but not the obligation, to attend bid and proposal evaluation sessions, subject to the provisions in the SSI Policy.
  - (c) If any other delivery method for procurement is selected, the Parties shall agree on the procedures for evaluation and selection.
- 7.2.5 NICTD shall lead the preparation of all procurement documents (including technical specifications) and of all contracts pertaining to the Rail Projects. Copies of all executed contracts shall be provided to IFA and RDA upon execution, but in all events, no later than submission of any related Payment Request.
- 7.2.6 Unless RDA and IFA consent otherwise, NICTD shall include in all contracts pertaining to the Rail Projects (and shall require that all lower-tier subcontracts or subconsultant agreements thereto also include) the following provisions:
  - 7.2.6.1 to the fullest extent permitted by law, the contracting party shall defend, indemnify and hold harmless, NICTD, IFA, RDA, and the State (and each of their respective agents and employees), from and against third-party claims arising out of or resulting from the services provided under the contract;

- 7.2.6.2 all of the insurance policies (excepting workers compensation and professional liability) provided by the contracting party shall provide that NICTD, IFA, RDA and the State are named as additional insureds;
- 7.2.6.3 the contracting party shall provide IFA and/or RDA access to all Books and Records, and shall provide IFA and/or RDA copies of all Books and Records upon request, at the contracting party's cost and expense;
- 7.2.6.4 without RDA or IFA assuming any liability or obligation thereunder, RDA and the IFA are intended third party beneficiaries of the contract and are entitled to enforce its terms, consistent with third party beneficiary rights and on the same terms and conditions as NICTD:
- 7.2.6.5 such contracts may be assigned to IFA and/or RDA or their respective designees in the event IFA or RDA exercises its Construction Period Step-In Rights pursuant to Section 11.4 herein;
- 7.2.6.6 payments thereunder shall be subject to State appropriation and availability, and that to the extent a determination is made by RDA or IFA that funds are not appropriated or otherwise available to support continuation of performance, such contract shall be subject to being terminated for convenience by NICTD at no cost to NICTD, RDA and/or IFA; and
- 7.2.6.7 NICTD, RDA and/or IFA shall have the right to monitor construction through independent technical representative(s), who also shall have access to the Rail Projects and all of the Books and Records (any such representative shall be subject to any contractual confidentiality terms and the SSI Policy).
- 7.2.7 The Parties shall confer and coordinate regularly regarding and during the development of the procurement documents. RDA and IFA shall have an opportunity to review and comment on all NICTD proposed final drafts of procurement documents for Major Contracts and NICTD proposed final drafts of all other Major Contracts and shall have not less than 30 days (excepting in case of emergency or as otherwise agreed by the Parties) to review and approve such NICTD proposed final draft procurement documents and such NICTD proposed final draft contracts prior to issuance and/or distribution to third parties. NICTD shall advise RDA and IFA of its final recommendation of the preferred proposer or bidder prior to award, and such preferred proposer or bidder shall be subject to RDA and IFA written consent and approval. RDA and IFA's consent and approval decision as to the preferred proposer or bidder and as to the form of final contract to be entered therewith shall be conveyed to NICTD within 10 business days of NICTD's recommendation. NICTD shall not enter any proposed Major Contract until RDA and IFA consent and approve in writing. Once RDA and IFA approve such contract in writing, and provided the amount and timing of the cost of such contract is as stated in the applicable Cost Budget and the applicable Financing Plan, NICTD shall be able to rely upon that commitment as binding in entering into its contractual obligations, subject to other terms herein.

- 7.2.8 Even after FFGA Approval, NICTD will not issue a notice to proceed to the design-builder or contractor, as applicable, on the Rail Projects until such time as NICTD is permitted to draw funds on the FFGA. For clarity, if the right to draw on the FFGA Funds is restricted by the FTA until NICTD receives a future approval from FTA or satisfies some other condition precedent to payment, issuance of a notice to proceed to the design-builder or contractor on the Rail Projects shall be withheld until such restriction by FTA is satisfied. NICTD will not be prohibited from issuing a notice to proceed merely because of an FTA administrative/procedural process (as opposed to a stated condition precedent).
- 7.3 Attached as Exhibits 3A, Exhibit 3B, Exhibit 4A, and Exhibit 4B are the agreed Cost Budgets and the Project Schedules for the Rail Projects, each of which may be amended or modified only by written agreement of the Parties. NICTD shall be responsible to ensure that project development, design and construction shall be performed to meet the applicable Cost Budget and Project Schedule, as amended by written agreement of the Parties. The Parties agree that: (a) the Cost Budgets include the Project Contingency required as part of the FTA requirement for the FTA CIG Program; and (b) the Financing Plans are designed to provide funding for the Cost Budgets. Notwithstanding that the Financing Plans are designed to fund the Cost Budgets, NICTD agrees to work closely with IFA and RDA to complete the Rail Projects using only the Project Contingency provided in Section 7.3.2, including the Project Contingency approved by IFA and RDA, so as to minimize the amount of funding required to fund actual Project Costs.
  - 7.3.1 The Cost Budgets, Project Schedules and written job cost reports prepared by NICTD per Section 7.4.1 shall be reviewed by NICTD, RDA and IFA on a monthly basis.
  - 7.3.2 The following proposed changes are subject to the review and written approval of RDA and IFA:
    - (a) changes to the applicable Cost Budget for either Rail Project,
    - (b) any use of Project Contingency set forth in the applicable Cost Budget, whether allocated or unallocated, in excess of \$250,000 (per month in the aggregate, per Rail Project);
    - (c) any adjustment to the applicable Project Schedule that will extend either the Revenue Service Operations Date or the date of Substantial Completion, or extend any other major milestone date established in the Project Schedule, for, in each case, either Rail Project; and/or
      - (d) the election of any Alternate.

#### 7.4 Monthly Reporting.

7.4.1 NICTD shall report to RDA and IFA on a monthly basis in a written job cost report in a form to be agreed upon by the Parties, which reconciles the historic expenditures and realistic projections for the costs of and schedule for the Rail Projects against the applicable Cost Budget and Project Schedule.

- 7.4.2 The monthly reporting shall include, but not be limited to: (i) a report on all proposed, rejected and approved change orders; (ii) a projected Project Contingency drawdown curve, actual use of Project Contingency, and the amount of remaining Project Contingency; (iii) a list of deviations between costs included in and underlying the applicable Cost Budget and the current projected actual cost of the Rail Projects (including, without limitation, all actual, incurred and estimated Project Costs); (iv) a list of deviations between the applicable Project Schedule and the actual projected schedule for the work, including any failure to achieve milestones and the effect, if any, on the cost forecast; and (v) copies of the monthly report submitted to FTA, including the risk register and third party agreement matrix.
- 7.4.3 NICTD must adhere to the reporting process in order to receive funding for approved Project Costs. NICTD will also inform RDA and IFA promptly in writing of any additional material deviations in the projected actual Project Costs versus the applicable Cost Budget, and any additional material deviations in the projected schedule versus the applicable Project Schedule. In the event of actual or projected Project Cost overruns or schedule delay, NICTD shall, as part of its monthly report to RDA and IFA, report what actions NICTD is taking or will take to bring the actual projected Project Costs within the applicable Cost Budget and/or the actual projected schedule within the applicable Project Schedule, as applicable.
- 7.5 RDA and IFA have established an agreed-upon budget for RDA/IFA Project Costs which are included in the applicable Cost Budget. RDA/IFA Project Costs shall be reported and shared with NICTD on an ongoing basis in the same manner and time as NICTD costs are shared with RDA and IFA.
  - 7.5.1 To the extent permitted by FTA, up to \$1 Million of the Project Contingency in the applicable Cost Budget may be used to fund any overrun in the RDA/IFA Project Costs (\$500,000 allocated to each of RDA and IFA).
  - 7.5.2 Any overrun of costs incurred by RDA and/or IFA above such allocated Project Contingency shall be paid for by RDA or IFA, as applicable, with funds other than FFGA funds or bond proceeds, except as provided in Section 3.5.2.4.4 (*i.e.*, the costs are necessarily incurred with respect to the Rail Projects) and/or except to the extent such overruns relate to Enhanced Oversight, Construction Period Step-In Rights and/or Operations Step-In Rights. In the event of either such exception, FFGA funds (to the extent permitted by FTA), bond proceeds and/or any other Project funding may be used to cover such RDA/IFA Project Costs. Notwithstanding the foregoing, NICTD shall not be responsible for paying any RDA/IFA Project Costs including any overrun of costs incurred by RDA and/or IFA, except to the extent such RDA/IFA Project Costs are paid from the NICTD Excess Cost Contribution, in accordance with Section 7.8, provided however that NICTD shall be liable for any such RDA/IFA Project Cost overruns that are caused by NICTD's failure to adhere to this Governance Agreement or its willful misconduct or gross negligence.

- 7.6 NICTD shall review, approve or reject, as appropriate, individual invoices submitted in accordance with the respective contract terms and NICTD's contractual payment obligations.
- 7.7 NICTD or NICTD's retained construction manager/technical representative/owners representative shall perform construction observation/inspection of the Rail Projects, verify contractor invoices, and provide such reports as reasonably requested by RDA or the IFA on thirty (30) days' advance notice.
- 7.8 Except as provided in Section 7.5 with respect to RDA/IFA Project Costs, or otherwise approved by IFA and RDA in writing, NICTD shall be responsible for the costs of the Rail Projects in excess of the applicable Cost Budget up to the amount of the Maximum NICTD Excess Cost Contribution. In addition, with respect to WL Delay Costs, NICTD shall be responsible for the NICTD West Lake Delay Cost Contribution, and, with respect to Double Track Delay Costs, NICTD shall be responsible for the NICTD Double Track Delay Cost Contribution. All costs, to the extent permitted by law, shall be submitted as a part of the overall Project Costs for potential federal grant reimbursement. In the event IFA or RDA has exercised Construction Period Step-In Rights, NICTD's obligation to provide the Maximum NICTD Excess Cost Contribution, the NICTD West Lake Delay Cost Contribution and the NICTD Double Track Delay Cost Contribution shall be as described in Section 11.4.2.
  - 7.8.1 In the event Excess Construction Costs are incurred, upon direction from IFA and RDA from time to time, in addition to paying the NICTD WL Delay Cost Contribution and/or the NICTD Double Track Delay Cost Contribution, as applicable, as required by Sections 7.8.3 and 7.8.4, NICTD agrees to make the NICTD Excess Cost Contribution by providing financing for such Excess Construction Costs, up to the Maximum NICTD Excess Cost Contribution. NICTD may choose to fund Excess Construction Costs from the payment of cash in lieu of funding the NICTD Excess Cost Contribution with the proceeds of bonds or a loan at the time of such direction.
  - 7.8.2 NICTD will not allocate more than \$30,000,000 of the Maximum NICTD Excess Cost Contribution toward the West Lake Corridor Project, nor more than \$15,000,000 of the Maximum NICTD Excess Cost Contribution toward the Double Track Project, without the written consent of IFA and RDA.
  - 7.8.3 NICTD's responsibility to pay for Excess Construction Costs of the West Lake Corridor Project shall consist of: (i) (a) its responsibility to pay for all WL Delay Costs as they are incurred to the extent the West Lake Corridor Project Cost Budget has been exceeded, and (b) to the extent WL Delay Costs were Advanced WL Delay Costs, its responsibility to pay for Excess Construction Costs of the West Lake Corridor Project that are not WL Delay Costs up to the amount of the Advanced WL Delay Costs, and (ii) its responsibility to pay for Excess Construction Costs of the West Lake Corridor Project that are not WL Delay Costs up to the West Lake Portion of the Maximum NICTD Excess Cost Contribution.
  - 7.8.4 NICTD's responsibility to pay for Excess Construction Costs of the Double Track Project shall consist of: (i) (a) its responsibility to pay for all Double Track

Delay Costs as they are incurred to the extent the Double Track Project Cost Budget has been exceeded, and (b) to the extent Double Track Delay Costs were Advanced Double Track Delay Costs, its responsibility to pay for Excess Construction Costs of the Double Track Project that are not Double Track Delay Costs up to the amount of the Advanced Double Track Delay Costs, and (ii) its responsibility to pay for Excess Construction Costs of the Double Track Project that are not Double Track Delay Costs up to the Double Track Portion of the Maximum NICTD Excess Cost Contribution.

- In the event there are Excess Construction Costs on the West Lake Corridor Project, to the extent NICTD is current on its obligations to pay WL Delay Costs as required by Section 7.8.3(i)(a) and after using all of NICTD's obligations to pay Excess Construction Costs that are not WL Delay Costs as required in Section 7.8.3(i)(b) and Section 7.8.3(ii), RDA agrees to contribute additional funds toward such Excess Construction Costs, in an amount not to exceed the RDA Excess Cost Contribution, and in accordance with the procedures set forth in the Financing Plan (Exhibit 5B). For clarity, NICTD is obligated to pay continuing WL Delay Costs as they are incurred as required in Section 7.8.3(i)(a) once the Cost Budget for the West Lake Corridor Project has been exceeded. In addition with respect to Excess Construction Costs of the West Lake Corridor Project that are not WL Delay Costs the priority for payment of such Excess Construction Costs shall be as follows: the portion of the NICTD West Lake Delay Cost Contribution described in Section 7.8.3(i)(b) hereof, if any, up to the full amount of Advanced WL Delay Costs shall be applied first to any such Excess Construction Costs of the West Lake Corridor Project, then the West Lake Portion of the Maximum NICTD Excess Cost Contribution will be applied to satisfy the next \$30,000,000 of West Lake Corridor Project Excess Construction Costs and lastly the RDA Excess Cost Contribution would then be used for the next \$30,000,000 of West Lake Corridor Project Excess Construction Costs. The RDA Excess Cost Contribution shall be used only for Excess Construction Costs that are not WL Delay Costs.
- 7.8.6 Subject to the Maximum NICTD Excess Cost Contribution, IFA and RDA may request that additional Excess Construction Costs be funded by NICTD at any time and from time to time until the Rail Projects are constructed and operational. The Parties agree that the first dollar in excess of each applicable Cost Budget as of the Effective Date shall be NICTD's responsibility to fund. Notwithstanding the foregoing, NICTD's obligation to fund Excess Construction Costs is limited to and shall be no more than the sum of the Maximum NICTD Excess Cost Contribution with respect to Excess Construction Costs, the NICTD West Lake Delay Cost Contribution with respect to WL Delay Costs and the NICTD Double Track Delay Cost Contribution with respect to Double Track Delay Costs in the manner provided in Sections 7.8.3 and 7.8.4.
- 7.8.7 The proceeds of the NICTD Excess Cost Contribution or any cash paid in lieu thereof, the NICTD West Lake Delay Cost Contribution, the NICTD Double Track Cost Contribution and the NICTD Double Track Delay Cost Contribution shall be deposited in the Revenue Trust Fund in the same manner as IFA bond proceeds, local contributions and FTA grant funds are deposited. The West Lake Portion of the NICTD Excess Cost Contribution shall be used toward West Lake Corridor Project Excess Construction Costs prior to any use of the RDA Excess Cost Contribution and prior to any

action by RDA or IFA to cooperate with NICTD to seek additional funding or additional revenues.

- 7.8.8 If additional funds are required to finance any Excess Construction Costs, other than WL Delay Costs and Double Track Delay Costs, after the total available funds from the Maximum NICTD Excess Cost Contribution and the RDA Excess Cost Contribution has been funded, the Parties shall seek additional funding or additional revenues through the Indiana General Assembly or from other sources.
- 7.8.9 On the later to occur of the Post-Acceptance Disbursement Dates for the Rail Projects, any balance of the NICTD Excess Cost Contribution remaining in accounts within the Revenue Trust Fund shall be released to NICTD. On the Post-Acceptance Disbursement Date with respect to the Double Track Project, any balance of the NICTD Double Track Delay Cost Contribution and the NICTD Double Track Cost Contribution remaining in accounts within the Revenue Trust Fund shall be released to NICTD. On the Post-Acceptance Disbursement Date with respect to the West Lake Corridor Project, any balance of the NICTD West Lake Delay Cost Contribution remaining in accounts within the Revenue Trust Fund shall be released to NICTD.
- 7.9 Any contribution NICTD provided on or before FFGA Approval shall be in addition to the Maximum NICTD Excess Cost Contribution, the NICTD West Lake Delay Cost Contribution, the NICTD Double Track Cost Contribution and the NICTD Double Track Delay Cost Contribution.
- 7.10 If the actual costs for completing the Double Track Project exceed the Cost Budget allocable to that project, NICTD and RDA shall certify such deficit, and the Parties shall jointly seek to enforce the contributions required from the entities described in IC 36-7.5-4.5-16(d)(5). Such contributions shall not constitute any part of the Maximum NICTD Excess Cost Contribution, the NICTD West Lake Delay Cost Contribution or the NICTD Double Track Delay Cost Contribution.
- 7.11 In the event NICTD recovers any liquidated damages or actual damages pursuant to Section 9.7 of the WL Design-Build Contract, to the extent that any such damages are equal to or less than the amounts set forth in Section 9.7.1.3(a) and (c) of the WL Design-Build Contract, NICTD shall transfer such damages to IFA, which will deposit such moneys in the IFA RDA Available Revenue Account of the Debt Service Fund established pursuant to the Revenue Trust Fund Agreement. Thereupon, IFA shall use such moneys to pay debt service on any IFA Bonds issued to finance the West Lake Corridor Project in accordance with the Revenue Trust Fund Agreement.
- 7.12 NICTD shall provide all documents and reasonable requests for information requested by RDA and IFA, including obtaining or requesting such documents or information from its contractors and consultants. RDA and IFA shall make such document and information requests to NICTD in writing and NICTD shall be provided a reasonable period of time in which to furnish or respond to such requests.

- 7.13 NICTD shall provide RDA and IFA unfettered access to all Books and Records and shall provide copies of all Books and Records requested by RDA and the IFA, subject to the SSI Policy.
- 7.14 RDA and IFA each retain the right to perform observation of construction by their respective independent technical representative for IFA's and RDA's respective benefit only.
- 7.15 NICTD shall provide the NICTD Double Track Cost Contribution in accordance with the Financing Plan for the Double Track Project (Exhibit 5A). Notwithstanding anything contained herein, the parties intend for the NICTD Double Track Cost Contribution to be used at IFA's discretion for the Double Track Project, for up to the final \$10,681,502 of Project Costs of the Double Track Project within the Double Track Project Cost Budget, and only after all other available funding sources provided for in the Financing Plan have been exhausted. The NICTD Double Track Cost Contribution shall not be applied to Excess Construction Costs. On the Post-Acceptance Disbursement Date with respect to the Double Track Project, NICTD will be released from any further obligation to advance any portion of the NICTD Double Track Cost Contribution not advanced prior to such date.
  - 7.15.1 In the event the Double Track Project achieves Final Completion and there are remaining funds within the Double Track Project Cost Budget, NICTD will be reimbursed for the amount of the NICTD Double Track Cost Contribution advanced from those remaining funds on the Post-Acceptance Disbursement Date for the Double Track Project.
  - 7.15.2 In the event additional federal funds become available for the Double Track Project prior to the provision of the full amount of the NICTD Double Track Cost Contribution by NICTD, the remaining portion of the NICTD Double Track Cost Contribution will be released, up to the amount such additional federal funds exceed any increase in the Double Track Project Cost Budget at that time when compared to the Double Track Project Cost Budget as of the Effective Date.
  - 7.15.3 In the event additional federal funds become available for the Double Track Project and any portion of the NICTD Double Track Cost Contribution has been advanced and the remaining portion of the NICTD Double Track Cost Contribution has been released in accordance with Section 7.15.2 above, NICTD shall be reimbursed up to the amount of the NICTD Double Track Cost Contribution advanced, up to the amount by which such additional federal funds exceed any increase to the Double Track Project Cost Budget at that time when compared to the Double Track Project Cost Budget as of the Effective Date.
  - 7.15.4 The NICTD Double Track Cost Contribution may be provided through the issuance of bonds by NICTD, through the provision of cash on hand of NICTD, or through some combination thereof.
  - 7.15.5 If other sources of non-federal funds become available for the Double Track Project, the funding of the Double Track Project Cost Budget will be recalculated

and the priority of sources will follow the process as determined in the Financing Plan for the Double Track Project.

## **ARTICLE VIII.**

#### **OPERATIONS AND MAINTENANCE**

- 8.1 NICTD shall operate and maintain the South Shore Line during construction and after completion of the Rail Projects.
- 8.2 Except as provided herein, NICTD shall be responsible for payment of all O&M costs of the South Shore Line, including during construction of the Rail Projects and after each of the Rail Projects are completed. Except as provided in <u>Section 8.4</u>, RDA shall have no responsibility for the costs of O&M of the Rail Projects.
- 8.3 NICTD shall pay O&M costs and rolling stock acquisition costs on the Double Track Project.
  - 8.3.1 NICTD shall have the ability to seek reimbursement for a portion of O&M and rolling stock acquisition costs on the Double Track Project from the FTA through federal formula funds, provided that any such request or reimbursement has no impact on the grant monies being pursued for reimbursement from FTA for capital costs for the Rail Projects.
  - 8.3.2 RDA and IFA shall have no responsibility to pay O&M or rolling stock acquisition costs for the Double Track Project, except the Parties shall work cooperatively to seek funding for O&M costs that NICTD is unable to fund or finance as provided in Article 6, and such costs shall not be treated as a cost of the Rail Projects under the Cost Budget.
- 8.4 NICTD shall pay O&M costs on the West Lake Corridor Project, subject to the further provisions in this <u>Section 8.4</u>, which shall govern in the event of a West Lake Operating Deficit.
  - 8.4.1 IC 6-3.1-20-7(d) provides that the RDA will receive certain Tax Credit Dollars, in an amount as determined by the Indiana Department of Revenue, to establish or improve public mass rail transportation systems in Lake County. Upon the execution of this Governance Agreement, the RDA shall deposit all Tax Credit Dollars in its possession into the Escrow Fund and shall thereafter upon receipt, deposit all Tax Credit Dollars into the Escrow Fund until Equivalent Alternative Funding is secured pursuant to Section 8.4.5 or WL Deficit Alternative Funding is secured pursuant to Section 8.4.6.
  - 8.4.2 All Tax Credit Dollars received by the RDA shall be maintained in the Escrow Fund and not utilized by RDA for any purpose other than toward a West Lake Operating Deficit, except as provided in <u>Section 8.4.6</u>.
  - 8.4.3 Beginning with the second calendar quarter after completion of the West Lake Corridor Project, RDA shall provide to NICTD, on a quarterly basis upon submission

by NICTD to RDA of a West Lake Deficit Cost Submission, from the moneys held at that time in the Escrow Fund (if available) a sum equal to the amount of West Lake Operating Deficit for the preceding calendar quarter. The Parties agree that each West Lake Deficit Cost Submission shall include a completed West Lake Deficit Cost Submission Form attached as <a href="Exhibit 8">Exhibit 8</a>, accompanied by a completed form of <a href="Exhibit 6">Exhibit 6</a> certified by NICTD. NICTD also shall provide RDA a copy of any West Lake Corridor deficit information NICTD provides to the FTA. The West Lake Operating Deficit shall be subject to verification by RDA and/or IFA in their discretion.

- 8.4.4 To the extent there exists a Subsidy Shortfall, NICTD will retain the right to collect such Subsidy Shortfall from the Tax Credit Dollars remaining in the Escrow Fund, except to the extent funds in the Escrow Fund are unavailable as a result of the operation of Section 8.4.6, and to collect such Subsidy Shortfall from future Tax Credit Dollars deposited into the Escrow Funds, except as otherwise provided in Section 8.4.5.
- 8.4.5 Upon the request of RDA, the Parties agree to cooperate in seeking funding to subsidize the West Lake Operating Deficit from the General Assembly. If at any time Equivalent Alternative Funding from the General Assembly to address the West Lake Operating Deficit is secured in an amount for the next year that equals or exceeds the immediately prior four quarters of Tax Credit Dollars, whether received by or provided to any Party or any other entity, in each case in any manner that is dedicated to subsidize the West Lake Operating Deficit, the RDA shall have no further or continuing obligation, whether under Section 8.4 or otherwise, to deposit Tax Credit Dollars in the Escrow Fund or to otherwise provide Tax Credit Dollars for the West Lake Operating Deficit (other than funds on deposit in the Escrow Fund when such Equivalent Alternative Funding is secured, which shall continue to be held, except as provided in Section 8.4.6, in the Escrow Fund and disbursed to NICTD to cover any portion of the West Lake Operating Deficit not covered by the Equivalent Alternative Funding).
- 8.4.6 As of the last day of the first year in which either (i) WL Deficit Alternative Funding is secured, or (ii) the Tax Credit Dollars collected for that year pursuant to Section 8.4.1, is equal to or greater than that year's West Lake Operating Deficit, any moneys that remain in the Escrow Fund (after disbursement to NICTD for such year's West Lake Operating Deficit), from whatever source, shall be released to RDA upon direction of the RDA.
  - (a) In the event WL Deficit Alternative Funding has been secured per <u>Section 8.4.6(i)</u>, RDA shall have no further or continuing obligation, whether under this <u>Section 8.4</u> or otherwise, to provide Tax Credit Dollars for the West Lake Operating Deficit or to deposit such Tax Credit Dollars into the Escrow Fund.
  - (b) In the event Tax Credit Dollars collected for that year pursuant to <u>Section 8.4.1</u>, are equal to or greater than that year's West Lake Operating Deficit per <u>Section 8.4.6(ii)</u>, Tax Credit Dollars as received shall continue thereafter to be deposited into the Escrow Fund and made available to NICTD to cover any West Lake Operating Deficit in accordance with the terms of this <u>Section 8.4</u>, until either Equivalent Alternative Funding or WL Deficit Alternative Funding is secured, in

either event of which RDA shall have no further or continuing obligation, whether under this <u>Section 8.4</u> or otherwise, to provide Tax Credit Dollars for the West Lake Operating Deficit or to deposit such Tax Credit Dollars into the Escrow Fund.

- (c) In no event shall RDA be responsible for West Lake Operating Deficits beyond (i) the Tax Credit Dollars collected under this <u>Section 8.4</u>, and (ii) cooperating with NICTD and IFA to seek the Equivalent Alternative Funding as provided in <u>Section 8.4.5</u>.
- 8.4.7 For avoidance of doubt, RDA shall have no responsibility for the costs of operations and maintenance of the Rail Projects beyond Tax Credit Dollars and only to the extent provided in this Section 8.4.
- 8.4.8 The Escrow Fund Agreement shall provide that RDA and NICTD shall receive monthly account statements. Until such time as WL Deficit Alternative Funding has been secured, by August 15 of each year, RDA shall provide NICTD a statement specifying the Tax Credit Dollars received and deposited into the Escrow Fund and disbursed from the Escrow Fund that are attributable to the preceding State fiscal year.
- 8.5 IFA and RDA shall not be responsible for the cost of rehabilitating or acquiring railway cars for use as rolling stock on the West Lake Corridor Project except to the extent such costs are included in the Cost Budget.
- 8.6 Neither IFA nor RDA shall be responsible for any lifecycle costs of the South Shore Line or the Rail Projects.
- 8.7 In addition to the NICTD Operations Financial Report required pursuant to <u>Sections 6.1 and 6.3</u>, at the end of each calendar year, NICTD shall prepare and furnish to IFA, RDA and others to whom it is statutorily or contractually required, an Annual Report that includes a certification as to whether or not its completed annual commuter rail services have been in substantial conformance with its published services schedule ("Level of Service") and also provides NICTD's current financial position and plans with respect to the following.
  - 8.7.1 The custody, safeguarding, investment and application of all moneys received or to be received by NICTD;
  - 8.7.2 The establishment of funds, reserves and accounts, including those required or held under the Bond Resolution;
  - 8.7.3 The construction, improvement, maintenance, repair, operation and insurance of the South Shore Line, including the anticipated Level of Service;
  - 8.7.4 The rates of fees, tolls, rentals, or other charges to be collected for the use of the South Shore Line; and
    - 8.7.5 Debt Coverage and Revenue Coverage for such year.

- 8.8 Additionally, by the end of January of each calendar year, NICTD shall prepare and furnish to IFA and RDA a certified report for the operation and maintenance of the South Shore Line that includes (i) the information set forth in Section 8.7 for the next succeeding calendar year, and (ii) a copy of NICTD's five-year capital plan (collectively "Business Plan"). A copy of the proposed Business Plan shall be provided to IFA and RDA for their review and comment at least 30 days before approval by the NICTD Board of Trustees. The Business Plan shall be approved by the NICTD Board of Trustees annually before it is implemented. Upon approval by the NICTD Board of Trustees, NICTD shall furnish a copy of the final approved Business Plan to IFA, RDA and others to whom it is statutorily and contractually required. Failure of NICTD to provide the Business Plan as required herein, the Annual Report as required by Section 8.7, or the NICTD Financial Operations Report required by Sections 6.1 and 6.3 shall be a material breach of this Governance Agreement and constitutes an NICTD Operations Adverse Condition under Section 11.5. The Annual Report and/or any parts thereof shall be subject to verification by IFA and/or RDA in their discretion.
- 8.9 During the term of this Governance Agreement, if at any point in time during the calendar year or as part of its annual year-end review and reporting, NICTD determines (1) that NICTD is not operating the South Shore Line in substantial compliance with the Business Plan then in effect, (2) that NICTD is not maintaining Revenue Coverage of 1.12 as set out in Section 6.5, or (3) that NICTD is otherwise in breach of any material provision in this Governance Agreement or the Development Agreement (if entered), with respect to the operations and maintenance of the South Shore Line and/or the Leases, including NICTD being otherwise unable to maintain the South Shore Line for use and occupancy in the manner provided in the Leases, NICTD shall so advise IFA and RDA, in writing, within 30 days of NICTD's determination, with a NICTD Initial Notice.

# **ARTICLE IX.**

#### **INSURANCE AND BONDING**

- 9.1 NICTD shall develop, or shall engage an insurance consultant to develop, for the review and approval of IFA and RDA, an insurance program for the Rail Projects and for the South Shore Line after completion of each Rail Project. NICTD's proposed insurance program shall be delivered to IFA and RDA by no later than fifteen (15) days prior to the issuance of any Request for Proposal that is related to construction or other work to be performed on-site for either of the Rail Projects.
  - 9.1.1 The insurance program for each Rail Project prior to its completion shall provide insurance protection for the benefit of IFA, RDA, NICTD, contractors, subcontractors and other Rail Project participants. The program may be an owner controlled program (OCIP), a contractor controlled program (CCIP), or a comprehensive program of insurance requirements for the insurance to be provided by the contractors, subcontractors or other consultants working on or providing services with respect to the Rail Project to be complied with by each participant on the Rail Project and shall include insurance and coverage amount requirements for:
    - (a) Workers Compensation

- (b) General Liability, including completed operations and contractual liability coverage
  - (c) Employer's Liability
  - (d) Automobile Liability
  - (e) Contractor's Pollution
  - (f) Pollution Legal Liability
  - (g) Builder's Risk
  - (h) Professional Errors and Omissions
- (i) Delayed opening coverage, for protection regarding payments on the IFA Bonds
- 9.1.2 The insurance program for the South Shore Line after completion of a Rail Project and continuing for the term of this Governance Agreement shall provide insurance protection for the benefit of IFA, RDA and NICTD and shall include insurance coverage amount requirements for:
  - (a) Workers Compensation
  - (b) General Liability
  - (c) Employers Liability
  - (d) Automobile Liability
  - (e) Property Insurance
  - (f) Business Interruption Insurance
- 9.1.3 NICTD, IFA, RDA, the State, the Bond Trustee, and the Deposit Trustee shall be named as additional insureds on such policies (except Workers Compensation and Professional Errors and Omissions insurance policies.)
  - (a) IFA, RDA and the State shall have no liability for deductibles or self-insured retentions.
  - (b) Each insurance policy shall provide that the coverage is primary and non-contributory with respect to the named and additional insureds.
  - (c) Coverages for Builder's Risk, Delayed Opening, Business Interruption, and Property Insurance shall provide coverage for all Force Majeure events as may be commercially reasonable and available.

- (d) Each insurance policy shall provide a waiver of subrogation in favor of all additional insureds.
- 9.2 NICTD shall require a 100% performance and payment bond from the Design-Builder and each of the Double Track Contractors, with the exception of the Double Track Contractor responsible for DT 6.
  - 9.3 Costs of insurance and bonding shall be included in the Cost Budget.

## ARTICLE X.

## **DISPUTE RESOLUTION**

- 10.1 For disputes between and among the Parties regarding the Rail Projects, RDA shall coordinate with NICTD and IFA to compile objective data to consider and analyze, along with the potential outcomes and impacts of a given decision or dispute. The Parties hereto, together with their consultants, shall support this process, as necessary.
- 10.2 The Project Executives shall confer in good faith in an attempt to reach a consensus on disputes after considering the compiled data and other appropriate factors, including the Decision Making Framework Factors. To the extent permitted by law, the Parties agree to maintain as confidential the deliberations (which include written communications and other documents) contemplated herein.
- 10.3 If consensus cannot be reached by the Project Executives on disputes between and among the Parties regarding the Rail Projects or proposed changes to this Governance Agreement or Development Agreement (if entered), then the dispute, issue, or proposed change shall be decided by a majority vote among the Project Executives, taking into consideration the Decision Making Framework Factors.
- 10.4 Any such majority decision by the Project Executives may be appealed to the State Budget Director within 10 business days of the decision.
- 10.5 Additionally, the State Budget Director (or his or her delegate) has the power to veto or override even a majority decision by the Project Executives for issues that, in the State Budget Director's sole discretion, materially and negatively impacts the State.
- 10.6 Except as provided in <u>Sections 15.6, 15.7 and 15.8</u>, nothing herein is intended to operate as a waiver of any Party's rights under this Governance Agreement, the Development Agreement (if entered into) or such contractual, statutory, common law, and/or equitable rights and remedies as may otherwise exist by law.

## **ARTICLE XI.**

## ADVERSE CONDITIONS / ENHANCED OVERSIGHT/STEP-IN RIGHTS

- 11.1 In addition to the remedy set forth in <u>Section 15.6</u>, IFA and/or RDA, shall be entitled to Enhanced Oversight on the Rail Projects in such manner and to such level as they deem appropriate in the event one or more of the following Construction Adverse Conditions occurs:
  - 11.1.1 The Revenue Service Operations Date or the date of Substantial Completion for either Rail Project is projected to be later than the date set forth in the applicable Project Schedule (as amended by written agreement of the Parties) or any major milestone date established in the applicable Project Schedule is not achieved by the date set forth in the applicable Project Schedule;
  - 11.1.2 The costs for either Rail Project is projected to be more than the amounts set forth in the applicable Cost Budget (not including any amounts added by written amendment agreed by the Parties), unless such increase is due to a change in scope requested by RDA;
  - 11.1.3 NICTD breaches any material provision in this Governance Agreement, the Development Agreement (if entered), and/or a Lease, including, but not limited to: (i) failure of NICTD to fund the NICTD Excess Cost Contribution, the NICTD West Lake Delay Cost Contribution or the NICTD Double Track Delay Cost Contribution as required by Section 7.8; (ii) failure of NICTD to fund the NICTD Double Track Cost Contribution as required by Section 7.15; (iii) failure to maintain Revenue Coverage of 1.12 as required by Section 6.5; (iv) incurrence of debt contrary to the provisions of Section 6.4; or (v) failure to comply with the requirements of Section 7.2.8;
  - 11.1.4 The occurrence of any circumstance that has a material adverse effect on the applicable Financing Plan;
  - 11.1.5 NICTD fails to comply with a decision rendered or direction given by the State Budget Director;
  - 11.1.6 NICTD, or an entity under the control of NICTD, takes action that deviates from the originally planned scope of work that has a materially negative impact on one or more TDDs or the development thereof;
  - 11.1.7 NICTD, or an entity under the control of NICTD, fails to effectively prepare any FTA grant application;
  - 11.1.8 The occurrence of any circumstance(s) that subject IFA or RDA to material liabilities not contemplated in the applicable Financing Plan; or
  - 11.1.9 NICTD, or any entity under the control of NICTD, fails to materially comply with FTA grant requirements and/or FFGA requirements.

- 11.2 In the event of the occurrence of one or more of the above Construction Adverse Conditions, IFA and/or RDA may send a notice to NICTD of intent to exercise Enhanced Oversight with respect to one or more Construction Adverse Conditions (each, a "Construction Adverse Condition Notice"). If the Construction Adverse Condition is not cured within 30 calendar days of NICTD's receipt of such Construction Adverse Condition Notice, in addition to all of their other rights hereunder, RDA and/or IFA shall have the right (but not the obligation) to exercise Enhanced Oversight on the Rail Projects, which shall include the right of RDA, IFA, and/or its designee(s), to engage in one or more of the following Enhanced Oversight activities at RDA or IFA's discretion (provided FTA has no objection after consultation):
  - 11.2.1 review and approve all change orders, contracts (and amendments thereto), and third party agreements (and amendments thereto) prior to execution;
    - 11.2.2 review and approve any allocations of remaining Project Contingency;
  - 11.2.3 require NICTD to prepare additional cost containment measures, including investigation of value engineering options or project scope changes;
  - 11.2.4 review and make recommendations regarding staffing and/or sequencing of Work;
  - 11.2.5 require NICTD to prepare and provide additional reports, including: job cost reports, scheduling reports, and/or additional value engineering/scope reduction alternatives:
    - 11.2.6 participate in all Project progress meetings;
  - 11.2.7 require NICTD to assist IFA and RDA, in their discretion, in seeking additional funding for the Rail Projects, including, but not limited to, from the General Assembly; and
    - 11.2.8 propose reduction or change in the scope of either or both Rail Projects.
- 11.3 The inclusion of items in <u>Sections 11.2.1 11.2.8</u> or <u>Sections 11.8.1 11.8.8</u> shall be cumulative of any rights provided to NICTD, IFA or RDA under this Governance Agreement, the Development Agreement (if entered into), or the Leases.
- 11.4 In the event a Construction Adverse Condition noted in Section 11.1 above, occurring prior to completion of construction of either or both Rail Project/s, is not curable or is not cured within 90 calendar days after the Construction Adverse Condition Notice, in its discretion, the IFA, RDA or one of their respective designees has the right, but not the obligation (provided FTA has no objection after consultation), to exercise Construction Period Step-In Rights and assume all facets of the design and construction of the Rail Projects, including but not limited to all of the following: design, construction, FTA grant application, land acquisition, utility relocation, and environmental approvals, until completion of the construction of the Rail Project(s).

- 11.4.1 To the extent the IFA and/or RDA (or one of their respective designees) exercises its Construction Period Step-In Rights under this <u>Section 11.4</u>, the IFA and/or RDA (as applicable) shall assume all responsibility for the design and construction of the Rail Projects and all further design and construction responsibilities of NICTD shall be terminated.
- 11.4.2 In the event IFA or RDA exercises Construction Period Step-in Rights and NICTD is no longer in charge of the Rail Projects for the duration of the construction thereof, NICTD shall not be responsible for any costs in excess of the applicable Cost Budget that are incurred by RDA or IFA, except for the Maximum NICTD Excess Cost Contribution, the NICTD West Lake Delay Cost Contribution and the NICTD Double Track Delay Cost Contribution. Notwithstanding the foregoing, NICTD shall be liable for any such excess costs that are caused by NICTD's failure to adhere to this Governance Agreement or its willful misconduct or gross negligence occurring either before or after NICTD is no longer in charge of the Rail Projects, to the same extent as if RDA and/or IFA had not exercised Construction Period Step-In Rights. For the duration of such exercise of the Construction Period Step-In Rights, NICTD, RDA and IFA shall coordinate the construction and railroad operations so as to minimize interference with the construction of the Rail Projects or the operation of the South Shore Line, including by developing a reasonable operations plan, and shall cooperate to effectuate assignment or enforcement of rights under contracts with respect to the Rail Projects.
- 11.5 If at any point in time during the calendar year, or as part of the annual year-end review and reporting, until Termination of this Governance Agreement, IFA and/or RDA determine that any of the following conditions exist, which are NICTD Operations Adverse Conditions, IFA and/or RDA may so advise NICTD in writing with an IFA Initial Notice or RDA Initial Notice, as applicable:
  - 11.5.1 that NICTD is not operating the South Shore Line in substantial compliance with the Business Plan then in effect,
  - 11.5.2 that NICTD is not maintaining Revenue Coverage of 1.12 as set out in Section 6.5, or
  - 11.5.3 that NICTD is otherwise in breach of any material provision in this Governance Agreement or the Development Agreement (if entered) with respect to the operations and maintenance of the South Shore Line, and/or the Leases, including that NICTD is otherwise unable to maintain the South Shore Line for use and occupancy in the manner provided in the Leases. Such a determination must be based upon documented financial statements or other evidence which demonstrate NICTD's alleged failure to meet and maintain the items set forth in this <u>Section 11.5</u>.
- 11.6 Within 60 days of the date of the NICTD Initial Notice under <u>Section 8.9</u> or within 60 days of the date of receipt by NICTD of an IFA Initial Notice or RDA Initial Notice, NICTD shall submit a written Recovery Plan to IFA, RDA and FTA setting forth NICTD's plans and proposed steps to cure the NICTD Operations Adverse Condition(s).

- 11.6.1 Any distribution of the Recovery Plan by IFA, RDA or other parties shall remain subject to the NICTD SSI Policy.
- 11.6.2 The Recovery Plan is subject to the review and approval of IFA and RDA. Assuming no objections by FTA after IFA and RDA consultation with FTA, within 30 days after receiving the Recovery Plan, the IFA, RDA and FTA may consult with NICTD and propose changes or additions to the Recovery Plan, which, in their best judgment, may enhance the Recovery Plan and improve it ("Consultation Period").
- 11.6.3 Barring written directive not to do so from IFA, RDA or FTA during the Consultation Period, NICTD shall commence the preliminary steps to implement the Recovery Plan upon its being tendered to the IFA, RDA and FTA and while waiting for formal approval thereof by IFA and RDA, provided there are no objections to the Recovery Plan by FTA.
- 11.6.4 NICTD shall commence and diligently prosecute such Recovery Plan to completion for a period not to exceed 120 days following receipt of the written approval of the Recovery Plan by IFA and RDA, as may be amended, or if such Recovery Plan requires work to be done, actions to be taken, or conditions to be remedied, which by their nature cannot reasonably be done, taken or remedied within such 120-day period, such 120-day period shall be extended for so long as NICTD shall commence such remedy within such 120-day period and shall diligently and continuously prosecute the same to completion, for a period not to exceed an additional 30 days (for a maximum of 150 days), or such longer period as agreed to in writing by the IFA and RDA ("Operations Cure Period"). NICTD regularly (and no less than monthly) shall keep IFA, RDA and FTA informed of the cure efforts NICTD is undertaking during the Operations Cure Period.
- 11.7 A failure by NICTD to maintain the South Shore Line for use and availability in the manner provided in the Leases, unless such failure is due to a Force Majeure Event, shall be an Availability Breach.
  - 11.7.1 In the event of an Availability Breach, notice thereof may be provided by IFA and/or RDA.
  - 11.7.2 In the event of an Availability Breach Notice from either IFA or RDA, NICTD shall have the Availability Cure Period in which to cure the Availability Breach.
- 11.8 Beginning with the date of a NICTD Initial Notice, an IFA Initial Notice, an RDA Initial Notice, and/or an Availability Breach Notice, the IFA, RDA and FTA, and its or their designees or consultants, subject to continued compliance with the NICTD SSI Policy, may monitor and observe the actions of NICTD in implementing the Recovery Plan or other efforts to cure (to the extent applicable) and to engage in one or more of the following activities:
  - 11.8.1 Review all NICTD contracts and proposed amendments to contracts prior to execution;
  - 11.8.2 Review and make recommendations to undertake cost containment measures;

- 11.8.3 Review and make recommendations regarding staffing, sequencing and/or performance of operations and maintenance;
- 11.8.4 Require NICTD to prepare and provide additional periodic reports, including, but not limited to, cost reports, scheduling reports, and/or status and progress reports on current or future operations and maintenance plans;
- 11.8.5 Require NICTD, consistent with applicable law, to pursue steps to increase revenues and/or cut expenses to the extent needed to cure a failure to achieve the Revenue Coverage of 1.12;
- 11.8.6 Prohibit NICTD from incurring any debt unless such debt is part of the Recovery Plan;
- 11.8.7 Review and approve any capital projects, other than those required by FTA;
  - 11.8.8 Participate in regular Recovery Plan scheduled meetings.
- 11.9 IFA may provide an Operations Step-In Notice in the event of any of the following: (i) at the conclusion of the Availability Cure Period, if NICTD has failed to cure the Availability Breach; or (ii) at the conclusion of the Operations Cure Period, if NICTD has failed to cure the NICTD Operations Adverse Condition(s) described in the NICTD Initial Notice, RDA Initial Notice and/or IFA Initial Notice, has failed to substantially complete and implement the Recovery Plan, or has failed to demonstrate its continued ability to maintain and operate the South Shore Line as statutorily required for its essential purpose as a commuter transportation rail line.
  - 11.9.1 After consultation with RDA and FTA, in addition to other remedies available under this Governance Agreement, IFA shall have the right to exercise the following Operations Step-In Rights effective immediately after sending to NICTD an Operations Step-In Notice: to assume or engage an operator acceptable to FTA of IFA's choosing or to take over all facets of the operation and maintenance of the South Shore Line for the Operations Step-In Period by assignment or other means.
  - 11.9.2 In the event IFA exercises its Operations Step-in Rights for operation and maintenance of the South Shore Line as provided herein, NICTD shall cooperate in all respects therewith, including but not limited to the NICTD Board of Trustees transferring or making available to the IFA, or any operator designated by IFA, all NICTD assets of the South Shore Line during the Operations Step-In Period.
  - 11.9.3 If prior to the expiration of the Operations Cure Period or the Availability Cure Period, as applicable, NICTD has cured all of the NICTD Operations Adverse Conditions set forth in Section 11.5 above, or as applicable, has cured the Availability Breach, IFA shall not be entitled to exercise any Operations Step-In Rights.

#### ARTICLE XII.

#### **RESTORATION**

12.1 Subject to the RDA-NICTD Operating Leases, if at any point during the term of this Governance Agreement, any part of the Rail Projects or the South Shore Line are destroyed or damaged by casualty or other cause of any kind or nature, NICTD shall undertake the repair or restoration using available insurance proceeds, to the extent that there is insurance coverage available, and other sources of funds available to NICTD.

## **ARTICLE XIII.**

# **TERMINATION**

- 13.1 Except for the Draw Down Bonds, once the IFA Bonds or any other similar financing by IFA or RDA are issued with respect to the Rail Projects or either Rail Project pursuant to this Governance Agreement, this Governance Agreement shall remain in full force and effect for so long as any such IFA Bonds, similar financings or any refunding bonds remain outstanding. When all such IFA Bonds, similar financings, and any refunding bonds with respect to the Rail Projects are retired, this Governance Agreement shall terminate, except in the event IFA has exercised Operational Step-In Rights, in which case this Governance Agreement shall continue unless and until one of the following occurs: (a) the IFA agrees to terminate this Governance Agreement; or (b) upon the written request of NICTD to the State Budget Director, the State Budget Director determines, in its sole discretion, that this Governance Agreement may terminate.
- 13.2 In the event the FTA declares that either the Double Track Project or the West Lake Corridor Project is not eligible for FFGA Approval, the provisions of this Governance Agreement shall no longer apply with respect to ongoing or future obligations with respect to that ineligible Rail Project, subject to applicable law.

## **ARTICLE XIV.**

#### THIRD PARTY AGREEMENTS

- 14.1 NICTD shall be responsible for securing all necessary third party agreements for the Rail Projects.
  - 14.2 All prospective agreements with third parties shall provide that:
  - 14.2.1 without RDA or IFA assuming any liability or obligation thereunder, RDA and the IFA are intended third party beneficiaries thereof and are entitled to enforce their terms, consistent with third party beneficiary rights and on the same terms and conditions as NICTD.
  - 14.2.2 the contract may be assigned to IFA and/or RDA or their respective designees in the event of the exercise of Construction Period Step-In Rights and/or Operations Step-In Rights per Article 11;

- 14.2.3 (for all contracts that are to be funded or financed by RDA or IFA) payments thereunder shall be subject to State appropriation and availability, and to the extent a determination is made by RDA or IFA that funds are not appropriated or otherwise available to support continuation of performance, such contract shall be subject to being terminated for convenience by NICTD at no cost to NICTD, RDA and/or IFA, except for costs of work performed or services provided prior to termination.
- 14.2.4 the third party is required to provide RDA and IFA with access to all documents related to the Rail Projects, and such third party shall provide copies of all documents to RDA at RDA's request, or the IFA, at the IFA's request, at the third party's cost and expense.
- 14.3 For all prospective agreements with utilities, municipalities and governmental agencies, NICTD shall seek agreement that any rights of NICTD shall extend to NICTD's assignees or successors in interest.
- 14.4 NICTD shall provide RDA and IFA the following information with respect to any amendment to any existing or prospective third party agreement proposed to be entered into after the Effective Date:
  - 14.4.1 the plan for securing such amendment or agreement; and
  - 14.4.2 the projected cost thereof.
- 14.5 Such plan and projected cost shall be subject to the review and approval of RDA and IFA, which shall take into consideration, without limitation, the Decision Making Framework Factors and whether FTA consent or approval is needed. Once RDA and IFA approve such agreement or amendment in writing, and provided the amount and timing of the cost thereof is consistent with the agreed upon projections that underlie the applicable Cost Budget and the applicable Financing Plan, NICTD shall be able to rely upon that commitment as binding in entering into such agreement or amendment, subject to the other terms herein. Such RDA and IFA approval is separate and distinct from the Major Contracts review and approval process set forth in Section 7.2.7.
- 14.6 In the event that IFA and/or RDA exercise any of its Construction Period Step-In Rights or IFA exercises any of its Operations Step-In Rights under <u>Article 11</u>, NICTD shall enforce its rights in all third party agreements as requested by RDA and/or IFA, as applicable.

# **ARTICLE XV.**

## **GENERAL MATTERS**

- 15.1 <u>Governing Law and Venue</u>. This Governance Agreement shall be governed by and construed in accordance with the laws of the State. Any suit must be brought in the State Court in Marion County, Indiana.
- 15.2 <u>Confidentiality</u>. The Parties and their respective consultants and contractors shall maintain the confidentiality of confidential data or information provided by the Parties and shall

not disclose such information to third parties not similarly bound to a non-disclosure agreement, except as may be required by statute, ordinance, or order of court, or as authorized by the Party which provided the data or information. The Parties and their respective consultants and contractors shall provide notice to the other Parties of any request for such information.

## 15.3 Press Releases and Public Records Requests.

- 15.3.1 Neither NICTD nor RDA shall make press releases or written statements regarding the Rail Projects without the prior consent of the IFA, NICTD and RDA.
- 15.3.2 The Parties shall provide to each other copies of all public records requests received as to the Rail Projects promptly after receipt and shall develop a process for a coordinated and mutually agreed response.
- 15.4 <u>Taxes</u>. NICTD shall perform its obligations hereunder such that purchases and contracts are exempt from sales and use taxes to the maximum extent permitted by law.
- 15.5 <u>Third Party Claims</u>. NICTD shall bear the costs of all losses or liabilities arising from tort claims for personal injury or property damage asserted by third parties with respect to the Rail Projects that are not covered by insurance, the Indemnitors as provided in <u>Section 7.2.6.1</u>, or another third party, except in the event that IFA and/or RDA exercises Construction Period Step-In Rights and/or IFA exercises Operations Step-In Rights, under <u>Article 11</u>. In the event of such exercise, NICTD shall bear the costs of such claims that arise prior to the effective date of the step-in, and those that arise after the Rail Projects are transferred back to NICTD, if applicable. The party (IFA or RDA, as applicable) that exercises such step-in rights shall bear the costs of claims that arise during any such period while step-in rights are being exercised.
- <u>Indemnity</u>. Except as provided in this <u>Section 15.6</u>, NICTD shall release, protect, defend, indemnify and hold harmless the RDA, IFA, the Bond Trustee, and the Deposit Trustee from and against any and all claims, causes of action, suits, judgments, demands, and/or losses, in each case if asserted or incurred by or awarded to any third party, arising out of or relating to or resulting from NICTD's design, construction or operation of the Rail Projects or the South Shore Line. NICTD's indemnity obligation shall not extend to: (i) any third-party claim to the extent directly caused by the negligence, recklessness, willful misconduct, or breach of any material obligation under this Governance Agreement by IFA or RDA, or (ii) any claims, causes of action, suits, judgments, demands, and/or losses resulting from any action of the RDA or IFA during an Operations Step-In Period or resulting from any actions taken by the RDA or IFA after the exercise of any Construction Period Step-In Rights, provided however that nothing herein shall relieve NICTD of responsibility to pay up to the Maximum NICTD Excess Cost Contribution for Excess Construction Costs, the NICTD West Lake Delay Cost Contribution with respect to WL Delay Costs, the NICTD Double Track Cost Contribution with respect to Project Costs for the Double Track Project or the NICTD Double Track Delay Cost Contribution with respect to Double Track Delay Costs, irrespective of the exercise of Construction Period Step-In Rights. The RDA and IFA shall have no obligation to indemnify NICTD.
- 15.7 <u>Liability Among the Parties</u>. Notwithstanding <u>Section 10.6</u>, none of the Parties shall be liable to any of the other Parties for claims and/or actions (whether alleging negligence,

breach of contract, strict liability, warranty, breach of professional services, detrimental reliance, or otherwise) relating to: (i) design, construction, operation or maintenance of any portion of the Rail Projects; (ii) financing or funding of any portion of the Rail Projects; and/or (iii) any other obligations hereunder, and each Party expressly disclaims any and all express or implied representations or warranties with respect thereof, including any warranties of suitability or fitness for use. The limitation of liability provided herein shall not apply to the following:

- 15.7.1 damages to the extent covered and paid for by insurance;
- 15.7.2 NICTD's indemnity obligation under <u>Section 15.6</u>;
- 15.7.3 damages to the extent covered and paid for by an Indemnitor pursuant to an indemnity obligation described in <u>Section 7.2.6.1</u>; and
- 15.7.4 damages to the extent paid for by a responsible party (other than a Party to this Governance Agreement) pursuant to applicable federal and state environmental laws.
- 15.8 <u>Injunctive Relief</u>. Notwithstanding <u>Section 10.6</u>, the Parties are entitled to seek injunctive relief for specific performance of any obligation set forth in this Governance Agreement, provided: such injunctive relief or specific performance relief is timely sought so as to not result in prejudice to the other Parties; and the IFA and RDA are entitled to pursue any remedies needed to protect the payment of the IFA Bonds.
- 15.9 <u>Fare Policy</u>. NICTD is authorized to set fare policy. NICTD as operator will collect and deposit farebox and any other revenues, including taxes, into the system revenue account (or similar account), except in the event IFA exercises Operations Step-In Rights under <u>Section 11.9.1</u>, in which case IFA shall have the authority to set fare policy and right to collect farebox and any other revenues during such Operations Step-In Period.
- 15.10 Amendment and Assignment. This Governance Agreement may be amended, supplemented, or modified only by a written document executed by the Parties. Except as otherwise provided herein or pursuant to applicable law, neither this Governance Agreement nor any of the rights, duties, or obligations described herein may be assigned by any Party hereto without the prior express written consent of the other Parties, and such consent shall not be unreasonably withheld so long as such assignment is consistent with the purposes of this Governance Agreement.
- 15.11 <u>Notice to Parties</u>. Notices under this Governance Agreement shall be in writing and: (a) delivered personally; (b) sent by certified mail, return receipt requested; or (c) sent by recognized overnight mail or courier services, with delivery receipt requested, to the following addresses:

As to RDA: President and/or Chief Executive Officer

Northwest Indiana Regional Development Authority

9800 Connecticut Drive Crown Point, IN 46307 With a copy to: Office of the Indiana Attorney General

302 W. Washington Street, 5th Floor

Indianapolis, IN 46204

As to NICTD: President and/or Chief Executive Officer

Northern Indiana Commuter Transportation District

33 E. US Highway 12 Chesterton, IN 46304

With a copy to: Office of General Counsel

Northern Indiana Commuter Transportation District

33 E. US Highway 12 Chesterton, IN 46304

As to IFA: Public Finance Director of the State of Indiana

Indiana Finance Authority

One North Capitol Avenue, Suite 900

Indianapolis, IN 46204

With a copy to: General Counsel

Indiana Finance Authority

One North Capitol Avenue, Suite 900

Indianapolis, IN 46204

15.12 <u>Authority</u>. Each of the undersigned attests that he/she is the properly authorized representative, agent, member or officer of the Party indicated.

- 15.13 <u>Severability</u>. If any provision of this Governance Agreement shall be held invalid, illegal or unenforceable, the validity, legality or enforceability of the other provisions hereof shall not be affected thereby, and there shall be deemed substituted for the provision at issue a valid, legal and enforceable provision as similar as possible to the provision at issue. This provision shall not be interpreted to materially alter the relationships of the Parties as set forth in this Governance Agreement or materially affect the ability of the Parties to achieve the purpose of this Governance Agreement.
- 15.14 <u>No Third Party Beneficiaries</u>. This Governance Agreement is solely for the benefit of the Parties hereto, and to the extent provided herein, their respective directors, officers, employees, agents and representatives, and no provision in this Governance Agreement shall be deemed to confer upon other persons any remedy, claim, reimbursement, cause of action or other right.
- 15.15 <u>Limitation on Recourse</u>. No recourse shall be had for the payment or performance of any obligation or covenant in this Governance Agreement, or for any claim against a Party to this Governance Agreement, personally against any past, present or future director, member, officer, employee, agent or official of any of the Parties under any rule of law or equity, statute, or constitution or by the enforcement of any assessment or penalty or otherwise, and all such personal liability is hereby expressly waived.

- 15.16 <u>Counterparts</u>. This Governance Agreement may be signed in any number of counterparts, each of which is an original and all of which taken together form one single document. Signatures delivered by email in PDF format or facsimile shall be effective.
- 15.17 <u>Non-Waiver of Rights</u>. Except as provided in <u>Section 15.7</u>, the failure of a Party hereto at any time or times to require performance of any provision hereof shall in no manner affect its right at a later time to enforce the same. No waiver by a Party of any condition or any breach of any term, covenant, representation or warranty contained in this Governance Agreement shall be effective unless in writing, and no waiver in any one or more instances shall be deemed to be a further or continuing waiver of any such condition or breach in other instances.
- 15.18 <u>Successors and Assigns</u>. This Governance Agreement shall be binding upon and inure to the benefit of the Parties and each of their permitted successors, assigns and legal representatives.
- 15.19 <u>Survival</u>. The dispute resolution provisions in <u>Article 10</u> and the provisions contained in this Governance Agreement, which by their inherent character should survive Termination, shall survive Termination.
- 15.20 <u>Further Assurances</u>. In connection with the structure contemplated by this Governance Agreement, each Party shall execute and deliver such additional documents and instruments and perform such additional acts that may be necessary or appropriate to deliver the Rail Projects in the manner described herein and/or to further the intent of this Governance Agreement.
- 15.21 <u>Subject to Appropriation</u>. Except with respect to those funds that as a matter of law are not subject to appropriation by the General Assembly, the Parties acknowledge that the funds for the Financing Plans under this Governance Agreement are subject to biennial appropriation by the General Assembly and the availability of the amounts necessary to fund such appropriations. The obligation of IFA and RDA to fund does not constitute an indebtedness of the State or any political subdivision thereof within the meaning or application of any constitutional provision or limitation, nor does it constitute a pledge of the faith, credit or taxing power of the State or any political subdivision thereof within the meaning or application of any constitutional provision or limitation. IFA and RDA have no taxing power. No Party has the right to have taxes levied or compel appropriations by the General Assembly for any payment contemplated hereunder.
- Governance Agreement, the Second Amended Governance Agreement and this Governance Agreement. The Parties agree that: (a) the Original Governance Agreement governed the Parties from the Effective Date of the Original Governance Agreement until the Effective Date of the Amended Governance Agreement governed the Parties from the Effective Date of the Amended Governance Agreement until the Effective Date of the Second Amended Governance Agreement; (c) the Second Amended Governance Agreement governed the Parties from the Effective Date of the Second Amended Governance Agreement until the Effective Date; and (d) the Parties are governed by this Governance Agreement commencing on the Effective Date.

[remainder of page intentionally blank]

IN WITNESS WHEREOF, the Parties, intending to be legally bound, have executed this Governance Agreement as of the date first written above.

# NORTHWEST INDIANA REGIONAL **DEVELOPMENT AUTHORITY**

By: Sherri Ziller

Title: President and Chief Executive Officer

# NORTHERN INDIANA COMMUTER TRANSPORTATION DISTRICT

By: \_\_\_\_\_

Name: Michael Noland

Title: President

## INDIANA FINANCE AUTHORITY

By: \_\_\_\_\_\_Name: Dan Huge

Title: Public Finance Director of the State of

Indiana

# APPROVED AS TO FORM AND SUBSTANCE:

By:
Zachary Q. Jackson, State Budget Director
APPROVED AS TO FORM AND LEGALITY:
ATTORNEY GENERAL
By:
Name:
Title: